Genetic Susceptibility to Multiple Sclerosis: Interactions between Conserved Extended Haplotypes of the MHC and other Susceptibility Regions

Goodin DS* ¹, Khankhanian P ², Gourraud PA ^{1,3,4}, Vince N ^{3,4}

1. Department of Neurology, University of California, San Francisco, CA, USA.

2. Center for Neuro-engineering and Therapeutics, University of Pennsylvania, Philadelphia, PA, USA

3. Centre de Recherche en Transplantation et Immunologie UMR 1064, INSERM, Université de Nantes, Nantes, France.

4. Institut de Transplantation Urologie Néphrologie (ITUN), CHU Nantes, Nantes, France.

Address for Correspondence:

Douglas S. Goodin, MD Department of Neurology University of California, San Francisco UCSF MS Center 675 Nelson Rising Lane, Suite #221D San Francisco, CA 94158 Phone: (415) 514 2464 Fax: (415) 514 2470 E mail: douglas.goodin@ucsf.edu

Author Contributions

- DSG: Conceptualized and led the project, analyzed and assisted in the interpretation of the data, developed the susceptibility Model, and wrote the original draft of the manuscript.
- PK: Assisted in the critical interpretation of the data and review of the manuscript.
- PA: Assisted in the critical interpretation of the data and review of the manuscript.
- NV: Assisted in the analysis and critical interpretation of the data and the review of the manuscript.

Abstract

OBJECTIVE: To study the accumulation of MS-risk resulting from different combinations of MS-associated conserved-extended-haplotypes of the MHC and three non-MHC risk-loci nearby genes EOMES, ZFP36L1, CLEC16A.

BACKGROUND: Defining "genetic-susceptibility" as having a non-zero probability of developing MS, both theoretical considerations and epidemiological observations indicate that only 2.2–4.5% of northern-populations can possibly be "genetically-susceptible" to MS. Nevertheless, many haplotypes (both within the MHC and elsewhere) are unequivocally MS-associated and, yet, have population-frequencies of >20%. Such frequency-disparities underscore the complex-interactions that must occur between these "risk-haplotypes" and MS-susceptibility.

DESIGN/MEHTODS: The WTCCC dataset was statistically-phased at the *MHC* and at three other susceptibility-regions. Haplotypes were stratified by their impact on "MS-risk". MS-associations for different combinations of "risk-haplotypes" were assessed. The appropriateness of both additive and multiplicative risk-accumulation models was determined.

RESULTS: Combinations of different "risk-haplotypes" produced an MS-risk that was considerably closer to an additive model than a multiplicative model. Nevertheless, neither of these simple probability-models adequately accounted for the accumulation of disease-risk in MS at these four loci.

CONCLUSIONS: "Genetic-susceptibility" to MS seems to depend upon the exact state at each "risk-locus" and upon specific gene-gene combinations across loci. Moreover, "geneticsusceptibility" is both rare in the population and, yet, is a necessary condition for MS to develop in any individual. In this sense, MS is a "genetic" disease. Nevertheless although, "genetic-susceptibility" is a necessary condition for MS to develop, environmental factors (whatever these may be) and stochastic processes are also necessary determinants of whether a "genetically-susceptible" individual will actually get MS.

Author Summary

Defining a "genetically-susceptible" individual to be any person in the population who has any chance of developing multiple sclerosis (MS), we demonstrate that, at a theoretical level and using widely-accepted epidemiological observations, only 2.2-4.5% of individuals in northern populations can possibly be "genetically susceptible" to MS. Thus, more than 95.5% of individuals in these populations have no chance of getting MS, regardless of the environmental circumstances that they may experience.

Nevertheless, certain "susceptibility-haplotypes" (e.g., *HLA-DRB1*15:01~DQB1*06:02*) have a far greater carrier-frequency than 2.2-4.5%. Consequently, most carriers of these "susceptibility-haplotypes" have no chance of getting MS and, therefore, their "susceptibility" must arise from some combination of these haplotypes with other "susceptibility-haplotypes". By analyzing such combinatorial impacts at four susceptibility-loci, we found significant interactions both within and between the different "susceptibility-haplotypes", thereby confirming the relationship between "genetic-susceptibility" and specific gene-gene combinations.

The nature of "genetic-susceptibility" developed here is applicable to other complex genetic disorders. Indeed, any disease for which the MZ-twin concordance rate is substantially greater than the life-time risk in the general population, only a small fraction of the population can possibly be in the "genetically-susceptible" subset (i.e., have any chance of developing the disease).

Introduction

The nature of susceptibility to multiple sclerosis (MS) is quite complex and involves both environmental and genetic factors [1-4]. Recently, considerable progress has been made in our understanding of the basis for "genetic-susceptibility" in MS. Thus, to date, over 200 common risk variants (located in diverse autosomal genomic regions) have been identified as being MS-associated by genome-wide association screens (GWAS) using large arrays of single nucleotide polymorphisms (SNPs) scattered throughout the genome [5-14]. Despite this recent explosion in the number of identified MS-associated regions, however, the association of MS susceptibility with certain alleles of the human leukocyte antigens (*HLA*) inside the major histocompatibility complex (*MHC*) has been known for decades [11,15-22]. Also, the importance of these new observations to our understanding of "genetic-susceptibility" in MS is tempered by the fact that any single SNP is generally associated with more than one gene or with more than one allele of a single gene. Moreover, sometimes the presumptively associated (i.e., "candidate") genes are at a considerable genetic distance from the location of the SNP itself [13,14].

For example, we have recently identified an 11-SNP haplotype (a1), which spans 0.25 megabases (mb) of DNA surrounding the HLA-DRB1 gene on the short arm of chromosome 6, and which has the most significant association with MS of any SNP haplotype in the genome [23,24]. Moreover, 99% of these (a1) SNP haplotypes carry the HLA-DRB1*15:01~HLA-DQB1*06:02 haplotype and, conversely, 99% of these HLA-haplotypes carry the (a1) SNP haplotype. In the Welcome Trust Case Control Consortium (WTCCC) dataset, the odds ratio (OR) for an association the full HLA-DRB1*15:01~HLA-DQB1*06:02~a1 haplotype was 3.28 $(p << 10^{-300})$ and similar disease associations for portions of this haplotype have been consistently reported in many other studies from northern MS populations [11,15-22,25]. Nevertheless, despite this extremely close association, and despite the fact that many of these 11 SNPs, individually, are highly associated both with this particular HLA-haplotype and with MS, for none of these individual SNPs is this association exclusive [26]. Thus, each of these SNPs is also found in association with other *HLA*-haplotypes [24,26]. Consequently, even with the large number of SNPs now identified as being MS-associated [13,14], any such association can only be viewed as simply tagging a relatively large genomic region; it cannot be used with confidence to identify any specific gene or to implicate any specific allele with respect to its role in causing, or contributing to, a "genetic-susceptibility" for MS.

Using data from the WTCCC, we recently reported that the *MHC* region was largely composed of a relatively small collection of highly conserved extended haplotypes (CEHs),

stretching across all of the "classical" HLA genes (HLA-A, HLA-C, HLA-B, HLA-DRB1, and HLA-DOB1) – a distance spanning more than 2.7 mb of DNA [26]. As shown in *File S2* (Supplemental Fig D), this same basic population structure is also found in numerous other widely separated human populations around the world [25]. These CEHs seem to be under a strong selection pressure, presumably based upon favorable biological properties of the complete haplotype [26]. Lastly, this population structure is unlikely to be the result of a linkage disequilibrium caused by the founder effects of a small population migrating out of Africa and radiating throughout Eurasia and the Americas. Rather, the marked divergence of the CEH composition both among and between these different human groups, including Africans (File S2; *Tables S4a & S4b*), indicates that this population structure must be due to selection. Consequently, "genetic-susceptibility" to MS, at least in so far as it relates to the MHC, is not likely to be attributable to any specific HLA allele but, rather, seems to depend upon the nature of each CEH [26]. Nevertheless, because many CEHs seem to be selected simultaneously and because the exact composition of the selected CEHs seems to be so fluid between different populations, the actual fitness landscape for this selection must be extremely variable in space and/or time and the introduction of novel allelic combinations must occur quite frequently [26].

Indeed, in the mostly European WTCCC population, the most frequent (and, thus, the most highly selected) Class II haplotype is HLA-DRB1*15:01~HLA-DQB1*06:02~a1, which accounted for 12.4% of all Class II haplotypes present in the control population. Nevertheless, most (or all) of the CEHs, which contain this Class II HLA-haplotype (including those whose full CEH had only a single representation in the WTCCC), are associated with an increased MS-risk, although the magnitude of the association varies significantly among the different CEHs [25]. Moreover, some rare haplotypes, which include the Class II motif of HLA-DRB1*15:01~HLA-DOB1*06:02 but not (a1), seem not to carry any risk [26]. By contrast, haplotypes containing (a1), but not this Class II HLA-motif, still carry substantial risk [26]. For the Class II HLA-motif of HLA-DRB1*03:01~HLA-DQB1*02:01, this dependence on the nature of the full CEH was even more evident. Thus, carriers of HLA-DRB1*03:01~HLA-DQB1*02:01~a2 seem to have a disease risk that is either dominant or dose dependent whereas most carriers of HLA-DRB1*03:01~HLA-DOB1*02:01~a6, seem to have a disease risk that is recessive or "neutral" [26]. Nevertheless, at least one such haplotypes (i.e., HLA-A*24:02~C*07:01~HLA-B*08:01~HLA-DRB1*03:01~HLA-DOB1*02:01~a6) has either a dominant or a dose dependent disease risk [26]. These examples underscore the complex interactions that take place between the various MHC alleles/haplotypes and MS-risk.

In the present manuscript, we explore these relationships and interactions between the different disease-associated CEHs in the MHC region and other "risk" haplotypes elsewhere in

the genome, in order to shed light on the nature of "genetic-susceptibility" to MS. Before embarking, however, we develop two underlying theoretical considerations. First, we summarize what is currently known about the various epidemiological parameters, which are associated with "genetic-susceptibility" in MS, and, in particular, about those parameters potentially related to the role of the MHC and other loci in producing this susceptibility. Second, we consider the different risk models used in epidemiology to account for the accumulation of disease-risk, which is caused by the combination of one or more MS-risk factors in the same individual.

Epidemiological Parameters Associated with MS

Several epidemiological parameters, which are associated with MS pathogenesis, can be defined and estimated and the relationships between them established using published epidemiological data. The definitions for the model parameters are provided in Table 1. Thus, population parameters, which are directly observable in the population as a whole, can be used to estimate non-population parameters, which cannot be measured directly, but which are, nonetheless, of considerable theoretical interest (*File S1*). These estimates and these relationships, summarized here, are developed comprehensively in the *S1 File*. For example *P*(*MS*), which represents the life-time probability that an individual in the general population of onset-ages for MS and the increased mortality experienced by MS patients [27-33], and each of these methods provides a remarkably consistent estimate for *P*(*MS*) in northern populations, which is $\sim 0.3\%$.

Nevertheless, each of these methods estimates only the prevalence of "diagnosed" MS, which may underestimate the prevalence of "pathological" MS in the population (*S1 File*). For example, several autopsy studies [34-37] have reported that "pathological" MS in patients is found in approximately 0.1% of individuals who were "undiagnosed" (i.e., they were either minimally symptomatic or asymptomatic) during life. In addition, at present, many of these asymptomatic individuals can now be identified (*in vivo*) using modern imaging methods [38]. As a result, the actual life time-risk of MS in the population is likely to be somewhat higher that this 0.3% estimate, and possibly by as much as 50–100% (*File S1*).

Also, within the general population (*Z*), we can define a subset of so-called "geneticallysusceptible" individuals (*G*), such that every individual in this subset has a non-zero probability of developing MS (Table 1). All individuals who are not members of the (*G*) subset, therefore, are members of the so-called "non-susceptible" subset (*G*–).

In addition, we will define the term $\{P(MS | MZ_{MS})\}$ to represent the life-time probability of developing MS for an individual from a monozygotic (*MZ*) twin-ship, given the fact that their

identical co-twin either has or will develop MS (Table 1). This probability is estimated by the proband-wise concordance rate for *MZ* twins [39] and most epidemiological studies in northern populations (e.g., Table 2) report this proband-wise concordance rate to be approximately 25–30% [40-46]. However, *MZ*-twins, in addition to sharing their identical genotypes (*IG*), also share similar intrauterine and early post-natal environments. Therefore, we define the term $P(MS | IG_{MS})$ to represent the *MZ*-twin concordance rate, which has been adjusted to account for these environmental similarities. As developed in the *S1 File*, this adjustment can be made using the observed proband-wise concordance rates of siblings and fraternal twins. – i.e., siblings who share the same genetic relationship but are divergent in their intrauterine and early post-natal experiences [40,47-49].

We also define two other widely reported partitions of the general population. First, we designate those individuals who possess 1 or 2 copies of the Class II *HLA-DRB1*15:01~HLA-DQB1*06:02~a1* haplotype – i.e. the (H+) haplotype – as being members of the (H+) subset and those who possess 0 copies of this haplotype as being in the (H-) subset. Second, the partition consisting of women (F) and men (M) is considered. Finally, we consider P(E) the probability that (G)-subset members will experience environmental conditions sufficient to cause MS, given the prevailing environmental conditions of the time

The estimated values of, and the conditional relationship between, these different subsets is comprehensively developed in the *S1 File*. However, ten of the more notable conclusions are:

$$\begin{array}{l} 0.067 < P(MS \mid G) \leq 0.134 \\ 0.022 \leq P(G) < 0.045 \\ P(G \mid H+) << 0.2 \\ P(G \mid H+) > 1.7 * P(G \mid H-) \\ P(F \mid G) < 0.41 \\ P(MS \mid G, F) > 2.7 * P(MS \mid G, M) \\ P(MS \mid G, F, IG_{MS}) = 5.9 * P(MS \mid G, M, IG_{MS}) \\ P(MS \mid G, H+, IG_{MS}) \approx P(MS \mid G, H-, IG_{MS}) \\ P(E \mid G, M) = 0.83 \\ P(MS \mid G, M, E) << P(MS \mid G, F, E) << 1 \end{array}$$

Each of these ten statements is unequivocal based on available epidemiologic data in Canada (*S1 File*). Notably, only 2.2–4.5% of the general population is even capable of getting MS – an estimate that is independently, and consistently, supported by epidemiological data from many populations throughout the northern hemisphere (*see File S1; Table S1*). This indicates that MS is fundamentally a genetic disorder. Moreover, the distribution of penetrance values within the general population (*Z*) is bimodal [50-54] with the large majority of individuals having no

chance of getting MS and a small proportion – those individuals in the subset (G) – having a unique predisposition to MS (*File S1*).

Notable, also, is the fact that the vast majority of (H+)-carriers (<<20%) do not belong to the (G) subset. Therefore, at least with respect to the (H+) haplotype, "genetic-susceptibility" to MS must be the result of the combined effect of (H+) together with the effects of other genetic factors (*File S1*). By itself, the (H+) haplotype carries no MS-risk whatsoever. Finally, the impact of specific environmental events on the development of MS is also critical (*File S1*). Thus, in addition to being a genetic disease, MS is also an environmental disease. Both factors are necessary; neither alone is sufficient (*File S1*).

Relative Risk Models in MS

There are two basic epidemiological models for the accumulation of disease-risk (*see File S2*), which have been widely utilized – the so-called additive and multiplicative risk models [55-59]. Nevertheless, actual epidemiological circumstances often don't fall neatly into one model or the other. Indeed, the same basic probability model can be used to approximate either an additive or a multiplicative accumulation of risk [56]. The difference depends upon the definition of the term "no interaction" between the risk-factors [55-59]. In studies of the "genetic-susceptibility" to MS, multiplicative risk models have generally been utilized [60-62], although this choice may not be appropriate in all circumstances (*see File S2*).

Indeed, in practice, there are certain difficulties, which are encountered when trying to assess the appropriateness of either model. First, in a case-control studies (such as the WTCCC), because the incidence of the disease is not assessed (as it would be in a prospective cohort study), the actual *RRs* cannot be determined [63]. However, for a rare disease such as MS {e.g., where: $P(MS)\approx0.003$ }, the *ORs* and the *RRs* are almost identical [63] and, thus, can be used interchangeably.

Second, and more important, is the selection of an appropriate reference group for calculating the *RRs* (*File S2*). This choice will, necessarily, influence how well any observations fit into one or another of these risk models. As noted above and discussed further in the *S2 File*, the theoretical underpinnings for both the additive and multiplicative models arise from the same underlying probability assumptions [55-58], and are predicated on the notion that MS-risk for the different potential "risk-factors" is as great or greater than the "risk" in the reference group (*File S2*). This requires identifying the reference group with the lowest MS risk of any. Although the (*G*–) subset, by definition, has the lowest MS risk of any, this group (even if it could be identified) cannot be used as a reference because all *RRs* calculated with respect to it would, by definition, be either infinite or undefined. Indeed, the fact that, for MS, the (*G*–) subset is non-

9

empty (*see File S1*), indicates that both of these "risk models" are, at a theoretical level, invalid for characterizing the accumulation of disease risk with an increasing number of diseaseassociated "risk" factors. Nevertheless, using a different reference group – i.e., one containing, at least, some members of the (*G*) subset – could be used to evaluate (approximately) whether either of these two models fits with the available data. In this circumstance, any subgroup consisting of only members of the (*G*–) subset would have an *RR* of zero. The subgroup with lowest risk of any that we identified in the WTCCC data was the (*AP**) subset. Therefore, this group was used for the present analysis, despite the fact that a subgroup with an even smaller (non-zero) disease-risk seems likely to exist (*File S2*).

Results

The MHC. There were 146 CEHs in the HLA region that had 50 or more representations in the WTCCC dataset and these accounted for 48% of the total number (59,884) of CEHs present. Information on 45 of the CEHs, which were found in our previous study [24] to have some relationship to MS susceptibility, is provided in the *S2 File (Tables S2 & S3)*. Of these, only the CEHs (*c1*, *c2*, *c3*, and *c5*) had a sufficient number of observations to assess the MS-risk of either homozygous combinations or combinations with each other. Therefore, these MSassociated CEHs were divided into five groups: 1) (*H*+) CEHs (i.e., containing the *HLA*-*DRB1*15:01~HLA-DQB1*06:02~a1* haplotype, *S2 File; Table S2*); 2) other increased risk or "extended risk" (*ER*) CEHs (*c23*, *c27*, *c34*, *c46*, *c68*, *c81*, *c85*, *c96*, and *c107*) CEHs as shown in *S2 File (Table S3*); 3) decreased risk or "all protective" (*AP*) CEHs (*c5*, *c15*, *c18*, *c24*, *c30*, *c32*, *c51*, and *c73*), as shown in *S2 File (Table S3*); 4) the "zero" group (*0*) consisting of all those CEHs which did not belong to the (*H*+), (*ER*), or (*AP*) groups; and 5) the (*c1*) CEH by itself. Each of these groups of CEHs seemed to be segregating independently and, in the control group, frequencies for each of the different combinations were, statistically, at their Hardy-Weinberg expectations.

The *MHC* allele *HLA-A**02:01 has been previously reported to be protective [64]. Although the association between *HLA-A**02:01-positive status and MS was also found, in the WTCCC, to be "protective" relative to the (0,0) *MHC* genotype (OR=0.69; p<10⁻²⁹), evidence from *Tables S2 & S3* (*S2 File*) shows that any such association depends importantly upon the exact nature of the CEH on which this allele resides rather than upon the presence of the *HLA-A**02:01 allele itself. Thus, the grouping used here (i.e., the *AP* group) seems more appropriate than the use of this allele in isolation.

The subset of individuals who don't carry any (H+), (ER), or (AP) CEHs at the *MHC* is referred to as the (0,0) *MHC* genotype. In Tables 2 & 3, all *ORs* are presented relative to this

group. Each of the (*H*+) CEHs with 50 or more representations were significantly associated with MS-risk (*File S2*; *Table S2*), as were, collectively, (*H*+)-carrying CEHs with fewer than 50 representations in the WTCCC (Table 3). Moreover, also assessing, collectively, only those (*H*+)-carrying CEHs that a single representation in the WTCCC, the disease association is still statistically significant and of similar magnitude to other (*H*+)-carrying CEHs (i.e., *OR*=3.0; CI=2.7–3.4; p<10⁻¹⁰). Consequently, the (*H*+)-haplotype, by itself, seems to contribute to the disease susceptibility in an individual although, as shown in *File S2* (*Table S2*), the magnitude of this effect varies among different (*H*+)-carrying CEHs [25].

In addition, as in Table 4 (see legend), we defined different "risk" CEH combinations as: 1) "single copy risk" [1 copy of any (H+)-haplotype or any ER haplotype]; and: 2) "double copy risk" [2 copies of any (H^+) -haplotype, (c1), or any ER haplotype, or any combination of $\{(H^+) +$ ER, {(H+) + (cI)}, or {ER + (cI)}]. The different "protective" CEH combinations were defined similarly as: 1) "single copy protective" [1 copy of an AP haplotype]; and: 2) "double copy protective" [2 copies of an AP haplotype]. Considering all of these "risk" CEH combinations (relative to the (0,0) MHC genotype), the (H+)-haplotypes accounted for 81% of the risk haplotypes in the control population and for approximately the same percentage of this risk in both men and women (80% and 82% respectively). Moreover, the likelihood of men in the control population possessing such a risk-CEH combination (26%) was approximately the same as the likelihood in women (27%). Similarly, the likelihood of men in the control population possessing an AP CEH (9%) was approximately the same as the likelihood in women (8%). Nevertheless, the "single copy risk" of MS for (H^+) - and (ER)-haplotypes in women (OR=3.0; CI=2.8-3.2; $p<10^{-220}$) was significantly greater (z=2.4; p=0.009) than the same risk in men $(OR=2.6; CI=2.4-2.8; p<10^{-96})$. By contrast, the "double copy risk" of MS in women and men was about the same.

Similar to the (H+)-haplotypes, CEHs carrying the *HLA*-motifs:

	(A2)-haplotype:	HLA-DRB1*03:01~HLA-DQB1*02:01~a2
and:	(A14)-haplotype:	HLA-DRB1*13:03~HLA-DQB1*03:01~a14

were associated with a disease risk that was similar regardless of the underlying frequency of the different CEHs (Tables 2 & 3). However, the same was not true for CEHs carrying the *HLA*-motif:

which seemed to vary quite widely in their disease association depending upon the exact CEH composition (Table 3; *S2 File; Table S3*).

The impact on the phenotype of an individual in response to combining two CEHs into a single genotype is shown in Table 4. For example, as has been well described previously [11,15-22], combining two copies of the (H+)-haplotype in to a single genotype markedly and significantly increases the disease association (Table 4; Fig. 1). Nevertheless, not all (H+)-carrying haplotypes have the same disease association [26]. For example, the *OR* for single copy carriers of the (c2) CEH is significantly greater (z=3.4–4.8; p=10⁻³–10⁻⁶) than the *OR* for either single or double-copy carriers of the (c3) CEH.

Similarly, considering the *AP* group of CEHs (Table 4; Fig. 1), we found a significant dose-dependent response such that possessing 2 copies of an *AP* CEH is significantly more "protective" than possessing only a single copy and, in addition, the magnitude of these "protective" effects is similar to the disease-risk produced by (*H*+)-haplotypes (Table 4; Fig. 1). Moreover, having an *AP* CEH, or even just the (*c5*) CEH, significantly and substantially mitigates (z=2.1-5.2; $p=0.02-10^{-7}$) the disease risk produced by single copies of (*c2*), (*c3*), or, more generally, any (*H*+)-haplotype (Table 4; Fig. 1). A single copy of an "extended risk" CEH adds to the risk of a single copy of (*c2*), (*c3*), or any (*H*+)-haplotype, although it adds significantly less (z=2.5; p=0.006) than does a 2^{nd} copy of an (*H*+)-haplotype (Table 4; Fig 1). And, finally, the (*c1*) CEH acts in a recessive manner with little, if any, disease risk produced by a single copy (Table 4). Nevertheless, (and by contrast) a single copy of the (*c1*) haplotype adds significantly (z=2.5-6.0; $p=0.006-10^{-9}$) to the disease risk produced by single copies of (*c2*), (*c3*), or, more generally, of any (*H*+)-haplotype (Table 4; Fig. 1).

Figure 2 shows the impact of replacing one MHC haplotype with another in different genotypic contexts. For example, replacing an (θ)-haplotype with an (H+)-haplotype has a significantly greater impact when the companion is an (θ)-haplotype compared to when the companion is an (H+)-haplotype (Fig. 2). Thus, comparing the (θ ,H+) genotype with the (θ , θ) genotype had an odds ratio of: (OR_1 =3.0) whereas, comparing the (θ ,H+) genotype with the (H+,H+) genotype had an odds ratio of: (OR_2 =2.1). These two *ORs* were significantly different from each other (z=4.7) and had a ratio of: OR_1 / OR_2 =1.4; and: ln(1.4)=0.4.

By contrast, replacing an (θ)-haplotype with an (H+)-haplotype has a significantly smaller impact when the companion is an (θ)-haplotype compared to when the companion is an (c1)-haplotype (Fig. 2). Thus, comparing the (θ ,H+) genotype with the (θ , θ) genotype had an odds ratio of: (OR=3.0) whereas, comparing the (θ ,c1) genotype with the (H+,c1) genotype had an odds ratio of: (OR=3.7). These two ORs were significantly different from each other (z=-2.2) and had a ratio of: (OR_1 / OR_2 =0.8; and: $ln(\theta.8)$ =- $\theta.2$.

As can be appreciated from the figure, the impact of replacing one haplotype with another often depends considerably (and significantly) upon the exact nature of the companion haplotype, which, together with the haplotype being replaced, constitutes the *MHC* genotype (Fig. 2). This reflects the multiple haplotype-haplotype interactions that exist within the *MHC*. Indeed, if no such interactions were present, each of the comparisons provided in the figure would have an OR of $\sim 1.0 - i.e.$, ln(OR)=0 – and would be shaded in yellow (Fig 2).

The Non-*MHC* Loci. In the WTCCC data set, and as described previously [24], the (*d1*) haplotypes are 11-SNP haplotypes in Region #234, which consist of 185 different SNP combinations and, of which, 1,243 (2%) are the "risk" haplotype (01100000100); the (*d2*) haplotypes are 3-SNP haplotypes in Region #734, which consist of 7 different SNP combinations and, of which, 14,091 (23%) were the "risk" haplotype (111); and the (*d3*) haplotypes are 15-SNP haplotypes in Regions #814, #818 and #822, which consist of 210 different SNP combinations and, of which, 24,709 (41%) are the "risk" haplotype (00001000000000). The *ORs* for the various combinations of the non-*MHC* loci are shown in Table 5. The increase in disease susceptibility that results from combining susceptibility genotypes at these three non-*MHC* loci with MHC genotypes is quite different for the different MHC configurations (Fig 3). Thus, for example, the different combinations of these non-*MHC* "risk" genotypes (Fig.3). By contrast, for other "risk" genotypes such as (*AP*,*H*+) and (*ER*,*H*+) and for "protective" genotypes such as (*AP*,*0*) and (*AP*,*c1*), these other these non-*MHC* "risk" haplotypes seemed to contribute essentially nothing to the final risk (Fig. 3).

Additive vs. Multiplicative Risk. Combinations of the 3 non-*MHC* susceptibility regions, together with different genotypes at the MHC are presented in Figs. 4–7. In each of these Figures, the *ORs* are those derived from a comparison with (*AP,AP*) *MHC* genotype individuals as the reference. In all cases, the disease risk conferred by each genotype at each locus is estimated directly from the WTCCC observations (*see Methods*). The expectations from the additive and multiplicative risk-models are then compared to the actual observations (Figs. 4–7). In almost all cases, the additive model fits better with the actual observations than does the multiplicative model, especially as more "risk" loci are included in the combinations (Figs. 4–7). Nevertheless, neither model fits perfectly. When considering only *MHC* "risk" genotypes, for combinations of *MHC* genotypes whose disease-risk exceeds that of the (0,0) *MHC* genotype, the actual disease-risk observed is, in general, greater than predicted by the additive model (Fig 4). By contrast, considering also the other non-*MHC* "risk" genotypes, the observed disease-risk is generally less than predicted by the additive model (Fig 5). This effect is increased when more "risk" loci are included in the combinations (Figs. 6,7).

Discussion

The present findings provide considerable insight to the underpinnings of "geneticsusceptibility" to MS and indicate that this susceptibility is complex. At the *MHC*, there are multiple different CEHs that contribute to susceptibility in different ways. When referenced to the *MHC* genotype (0,0), certain groups of CEHs seem to affect disease risk in a manner that either increase or decrease disease-risk when combined into a single genotype. For example, the combination of 2 "risk" CEHs (*H*+ or *ER*) results in an increased disease risk compared to a single copy of a "risk" CEH alone (Tables 2 and 3, Fig. 1). Similarly, the combination to 2 "protective" CEHs ("protective" or "all protective"), results in a decreased disease risk compared to a single copy (Table 4; Fig. 1). Finally, combining a "risk" CEH together with a "protective" CEH results in an intermediate disease risk compared with having a single copy of either CEHtype alone (Table 4).

Nevertheless, there are exceptions to this general rule. Notably, when referenced to the (0,0) *MHC* genotype, a single copy of the (c1) CEH – the highest frequency CEH in both the WTCCC controls and other European populations [25,26] – is associated with a negligible, nonsignificant, disease-risk (Table 4; Fig. 1). By contrast, the disease-risk is substantially (and significantly) increased in the homozygous state (Table 4; Fig. 1). Such a pattern suggests that (c1) is acting in a recessive manner. Nevertheless, a single (c1) CEH increases disease risk when combined with "risk" CEHs but not with "protective" CEHs (Table 4). Thus, (c1) with an (H+) or an (ER) CEH resulted in a significantly increased disease risk compared to each CEH alone (Fig. 1). By contrast, the combination of (c1) and an (AP) CEH neither enhances nor to mitigates the effect of the "protective" CEH by itself (Fig.1).

Our findings have certain implications with respect to the appropriateness of the additive and multiplicative causal models for the accumulation of genetic risk. For appropriate OR or RRcomparisons, calculations, need to be made using the lowest, non-zero, MS-risk as a reference group. In the WTCCC, the (AP,AP) MHC genotype had the lowest risk (OR=0.13) relative to the (0,0) MHC genotype of any that we identified (Table 4; Fig.1). Therefore, this group was used to normalize the MHC haplotype risk effects. We show that all MHC genotypes, except (c1), are intermediate between the two causal models (Fig. 4). By contrast, for (c1,c1) and (c1,ER) genotypes, the observations exceed the expectations of both models (Fig. 4).

When the other non-*MHC* "risk" loci are included in the analysis, observations are closer to the additive model. Thus, the estimates from a multiplicative model exceed observations by 1-2 orders of magnitude (Figs. 5-7). As demonstrated previously for a different definition of the (*G*)

subset [3], the distribution of penetrance values in the general population (Z) is incompatible with a lognormal distribution [3] – i.e., the distribution expected for a multiplicative model. In the present iteration of the model, defining the set (G) to include all genotypes, which that have a non-zero expected penetrance, the bimodality of the distribution can be established with certainty (*File S1*). Consequently, based upon both theory and observation, a multiplicative model for the accumulation of genetic risk in MS is inappropriate.

The additive model, in general, performed better in these circumstances (Figs. 4-7). Nevertheless, it does not explain perfectly the accumulation of genetic risk in MS. First, (*c1*) CEH genotypes consistently exceed the additive expectations (Fig. 4). Second, effect of a given *MHC* haplotype is dependent on its companion *MHC* haplotype in a genotype (Fig. 2). Third, the effect of the 3 non-*MHC* "risk" haplotypes is not consistent across all *MHC* genotypes (Fig. 3). And fourth, when more loci are included in the analysis, the observations become increasingly less than what is predicted by the additive model (Figs. 5-7). Taken together, these lines of evidence indicate that the accumulation of genetic risk from these "susceptibility loci" is inconsistent with both an additive and a multiplicative model. Rather, the magnitude of any change in disease-risk associated with the inclusion of additional "susceptibility loci" seems to depend upon the exact state at each "risk-locus" and on the interaction across all loci. Such a conclusion is also anticipated on the basis of theoretical considerations (*File S1*).

The *MHC* is known to have a remarkable diversity [63]. In the WTCCC population, there were 29 *HLA-A* alleles, 29 *HLA-C* alleles, 55 *HLA-B* alleles, 35 *HLA-DRB1* alleles, and 16 *HLA-DQB1* alleles. Moreover, these alleles did not exist in isolation but, rather, as part of 10,078 unique CEHs, 810 of which accounted for 71% of all the CEHs present in the WTCCC dataset [26]. Also, even if some CEHs share common features, such as carrying the (*H*+)-haplotype, the degree of association with MS varies depending upon the exact CEH considered (*S2 File; Tables S2 & S3*). For example, both (*c2*) and (*c3*) CEHs carry the (*H*+)-haplotype, but their MS-association differed significantly (*z*=4.8; *p*<10⁻⁶). It might be tempting to attribute this difference to (*c3*) carrying the potentially "protective" *HLA-A**02:01 allele (*S2 File; Table S2*). However, other *HLA-A**02:01 and (*H*+) carrying CEHs (e.g., *c50, c58*, and *c139*) do not seem to be similarly protected (*S2 File; Table S2*). Finally, each identified CEH probably represents a diverse set of CEHs. Thus, because the 3 mb genomic region from *HLA-A* to *HLA-DQB1* is quite "gene-dense", each of the CEHs that we defined, almost certainly, represent groups of CEHs, which carry many other linked polymorphisms.

Although the non-*MHC* "risk" regions used for this analysis are likely to be less variable than the MHC, these regions span large amounts of DNA (200-680 kb) and they generally have hundreds of highly conserved SNP-haplotypes across each region. Moreover, despite the fact that

authors sometimes identify specific genes as being MS-associated [13,14], the truth is that we have no basis for deciding which gene or genes within a region are responsible for the association. We cannot exclude the possibility that, within these regions, as within the MHC, there might exist "risk" or "protective" alleles interacting with each other. If so, the likelihood that any simple probability model (either additive or multiplicative) will adequately describe genetic-susceptibility to MS seems quite remote.

However, such complexity fits well with the model of genetic-susceptibility presented in the *Introduction* and more fully developed in the *S1 File*. Thus, MS-susceptibility – i.e., membership in the subset (*G*) – seems to be confined to a small subset (~2.2– 4.5%) of the general population (*File S1; Table S1*) and, yet, this susceptibility is a prerequisite to getting MS, with members of the (*G*–) subset having no chance of getting MS, regardless of what environmental experiences they have (*S1 File*). Moreover, despite the fact that the Class II (*H*+)-haplotype is, by far, the strongest, and most significant, MS-associated genetic factor (p<<10⁻³⁰⁰) of any in the genome and has been known for over a half a century [11,15-22,26], only a tiny fraction of (*H*+) carriers are even "susceptible" to getting MS (*File S1*). This observation indicates that at least with respect to the (*H*+)-haplotype, "genetic-susceptibility" to MS requires the combined effects of different genes (*File S1*). The presence of (*H*+), by itself, does not increase disease-risk.

Nevertheless, despite the necessity of being a member of the (G) subset in order for a person to develop MS, environmental factors are also required. Indeed, once "geneticsusceptibility" is established in an individual, these environmental factors, together with certain stochastic processes, are entirely responsible for determining who does and who does not ultimately develop MS (File S1). Some of these causal environmental factors seem to occur in *utero* or, possibly, in the early post-natal period; others seem to occur during adolescence; and still others seem to occur later [50,51,65-67]. There is strong evidence that Epstein Barr viral infections (especially those associated with symptomatic infectious mononucleosis) are causally associated. There is also strong circumstantial evidence that Vitamin D deficiency is an important factor [50,51,65-67]. Other factors (e.g., smoking, obesity, and possibly other infections) may also play a role [50,51,65-67]. Regardless of the identity and role of each factor, however, it seems that, collectively, these environmental events (which currently occur as "population-wide" exposures) are major determinants of whether or not the disease will develop in a "susceptible" individual (*File S1*). For example, although the (H+)-haplotype, as noted, has the strongest association with MS of any, the possession of this haplotype seems only to make (G) subset membership more likely but does not seem to alter the likelihood of actually getting MS once (G) subset membership is established (Table 1: S1 File).

Similarly, the well-known gender-bias in MS prevalence seems to be largely explained by an increased responsiveness of "susceptible" women to the environmental factors involved in MS pathogenesis (*File S1*). Nevertheless, men are more likely to be members of the (*G*) subset than are women (*File S1*). Such a finding might seem surprising given the facts that 1) all of the ~200 MS-linked loci are on autosomal chromosomes [13,14]; 2) association studies specifically focused on the *X*-chromosome have not suggested the presence of any *X*-linked associated loci [7]; and, finally, 3) it is hard to rationalize how women could possibly be more or less likely to possess any specific autosomal genotype compared to men (*S1 File*). Indeed, in the WTCCC, women seemed to be equally likely to possess both the "risk" and the "protective" CEH combinations compared to men. Nevertheless, if (*G*) subset membership depends upon specific genetic combinations, and defining the subset (*G_a*) to represent the autosomal genotypes in the general population (*Z*), it is certainly plausible that men and women could be equally likely to possess each of its members and, yet, for any specific (*k*th) member with genotype (*G_{ak}*), it could well be the case that:

 $(G_{ak}, M) \in (G)$ and, that: $(G_{ak}, F) \notin (G)$

Such a circumstance would represent another example of "genetic-susceptibility" to MS requiring the combined effect of different genetic traits (*File S1*).

Moreover, from the epidemiological observations regarding changes in sex-ratio that have taken place over time [68], the response curves for "susceptible" men and women to increasing levels of environmental exposure can be derived quantitatively (*Supplemental Fig. C; File S1*) and it is clear from these response curves that women are, indeed, more responsive (probably physiologically) than men to the causal environmental factors involved in MS pathogenesis [3,4,68]. Moreover, this analysis strongly suggests that the relevant environmental exposures must be occurring currently at "population-wide" levels (*S1 File*). Such a conclusion is fully consistent with the same conclusion reached from observational studies in adopted individuals, in siblings and half-siblings raised together or apart, in conjugal couples, and in brothers and sisters of different birth order, which have generally indicated that MS-risk is unaffected by the childhood or other micro-environments [69–75].

By contrast, comparing the penetrance in the (H+) and (H-) subsets of *MZ*-twins, the fact that there is little difference in penetrance between the $(H+,G,IG_{MS})$ and $(H-,G,IG_{MS})$ subsets strongly suggests that there is also little difference in penetrance between the (H+,G) and (H-,G) subsets. Indeed, as demonstrated in *File S1*, despite the fact that the disease association for the (H+) subset is, by far, the strongest and most significant of any in the entire genome [11,15-22], this association is due mostly to the fact that: P(G | H+) >> PG | H-).

In the study of human genetics there has been a long-running debate between the socalled "common-disease, common variant" (*CDCV*) and the "common-disease, rare variant" (*CDRV*) hypotheses [76]. Nevertheless, with our improved genetic sophistication, it has become increasingly clear that, in different specific circumstances, either (or both, or neither) hypotheses could be operative [76]. In fact, our observations also support this notion. For example, on the one hand, all of the *MHC* CEH combinations, which impact MS-susceptibility, are quite rare. None has a population frequency in controls of more than 6.2% and the large majority of them have population frequencies well below 1% (*S2 File; Tables S2&S3*). On the other hand, considered collectively, those CEH combinations, which include the Class II (*H*+)-haplotype, have a WTCCC Control population frequency of 23%. Indeed, this particular haplotype-group is the most prevalent (and, therefore, the most highly selected) of all such Class II haplotype combinations in northern population [26].

Consequently, regardless of whether one considers these observations to be supportive of an association between MS-susceptibility and "common" or "rare" variants, the fact remains that, whether considered individually or collectively, the most prevalent, and therefore the most highly selected [26], CEHs are those that are also associated with the highest MS-risk (S2 File; Tables S2&S3). Thus, it is clear that these particular CEHs must come with both adaptive and deleterious consequences for the individual. Also, although the CEH composition differs markedly among long-separated human populations (File S2; Tables 4a & 4b), specific CEHs are still being strongly selected in each of them [26]. Consequently, the benefits of the adaptive features of these CEHs must outweigh the risk of any deleterious ones. Obviously, for circumstances, either in which the risk of MS is small or in which MS has little impact on an individual's eventual number of surviving children, even a modest advantage in favor of a specific CEH might still cause it to be selected. In this regard, a recent French study estimated that women with MS had 31% fewer children than their contemporary controls [77]. If this observation is correct, it suggests that there is a strong selective disadvantage to having MS. Therefore, the explanation for the benefits of these MS-associated CEHs outweighing the risks is likely to lie in an individual's low risk of MS rather than the disease having little impact on their fertility. Based on our observations, this seems likely to be the case. Thus, because natural selection can only select against those genotypes, which actually carry risk (relative to other genotypes), the fact that so few members of the "susceptible" (G) subset ever actually develop MS makes such a favorable tradeoff between adaptive and deleterious features considerably more likely to occur.

Our results also bear on the common notion that there is a considerable amount of "missing heritability" in both MS and other complex genetic disorders [78-80]. First, as discussed in *File S1*, much of the variability in MS expression (even among "genetically-susceptible"

18

individuals) is attributable to stochastic processes that are unrelated to either environmental or genetic factors. Second, MS expression is related to an interaction between the environmental and genetic factors involved in MS pathogenesis; neither alone are sufficient and both are necessary (*File S1*). Third, both theoretically (*File S1*) and observationally (Figs 2 & 3) specific gene-gene combinations are crucial determinants of "susceptibility" to MS – a circumstance which renders the common (additive) methods of estimating heritability unreliable [77]. And fourth, with over 200 independent MS-associated genetic regions [5-14], each potentially with more than one "susceptible state" (e.g., the MHC), there are so many possible combinations of states at these loci that, almost certainly, every person with MS possesses a unique combination. If, as indicated by our results, only a few of these combinations are members of the (*G*) subset, even of combinations that are similar to each other (*File S1 & File S2; Table S2*), then there are more than enough genetic associations identified already to account fully for membership in the (*G*) subset. Naturally, many more MS-associated loci may yet be identified in the future although their existence is not necessary (*File S1*).

Alternatively, if "missing heritability" is meant to imply only that our genetic model cannot predict accurately the occurrence of MS, then it is true that a substantial amount of the "heritability" remains unexplained. Indeed, the environmental factors, gene-gene combinations, gene-environment interactions, and stochastic factors, which underlie the development of MS in any individual, are poorly understood, thereby making any accurate prediction of MS occurrence, at present, impossible.

Finally, it is worth noting that the nature of "genetic susceptibility" developed in this manuscript is applicable to a wide range of other complex genetic disorders such as type-1 diabetes mellitus, Crohn's disease, ulcerative colitis, and rheumatoid arthritis. Indeed, base solely upon *Proposition #1 (S1 File)*, any disease for which the MZ-twin concordance rate is substantially greater than the life-time risk in the general population, only a small fraction of the population can possibly be in the "genetically susceptible" subset (i.e., have any chance of developing the disease). Moreover, any disease for which the MZ-twin concordance rate is substantially less than 100% must, in addition to "genetic susceptibility", include environmental factors, stochastic factors, or both in the causal pathway leading to the disease.

Materials & Methods

Ethics Statement

This research has been approved by the University of California, San Francisco's Institutional Review Board (IRB) has been conducted according to the principles expressed in the Declaration

of Helsinki.

Study Participants

<u>Wellcome Trust Case Control Consortium (WTCCC)</u>. This multinational study cohort consists of 18,872 controls and 11,376 cases with MS and has been described in detail previously [13,14]. However, SNP haplotype data was unavailable for 380 controls and 232 cases. Of the cases, 72.9% were women, the average age-of-onset was 33.1 years, and the mean Extended Disability Status Score (EDSS) was 3.7 [14]. The patients enrolled in this study (except for a few African Americans from the United States) were of European ancestry. The large majority (89%) of the cases had a relapsing-remitting onset [13]. The diagnosis of MS was made based upon internationally recognized criteria [81-83]. Control subjects were composed of healthy individuals with European ancestry [13]. The Ethical Committees or Institutional Review Boards at each of the participating centers approved the protocol and informed consent was obtained from each study participant. The WTCCC granted data access for this study.

Genotyping, and Quality Control

The genotyping methods and quality control for the WTCCC have been described in detail previously [13,14,16,18,19]. All genotyping was performed on the Illumina Infinium platform at the Wellcome Trust Sanger Institute. Case samples were genotyped using a customized Human660-Quad chip. Common controls were genotyped on a second customized Human1M-Duo chip (utilizing the same probes). After quality control, this provided data on 441,547 autosomal SNPs scattered throughout the genome in both MS patients and controls. The identities of the five HLA alleles in the MHC region (*A*, *C*, *B*, *DRB1* and *DQB1*) were determined for each participant by imputation using the HIBAG method [84].

Statistical Methods

<u>Phasing.</u> Both the phasing of alleles at each of five *HLA* loci (*HLA-A*, *HLA-C*, *HLA-B*, *HLA-DRB1* and *HLA-DQB1*) and the phasing of the SNP-haplotypes surrounding the Class II region of the *DRB1* gene were accomplished using previously published probabilistic phasing algorithms [23,24,85,86]. SNP-haplotypes from 3 of the 102 non-MHC genomic regions, which had been identified previously as being significantly MS-associated, were also included in our analysis [24]. In our previous report, the MS-associated SNP haplotypes were numbered (arbitrarily) from 1 to 932. These three particular regions (arbitrarily labeled *d1*, *d2*, and *d3*) were selected based on their having a "risk" SNP-haplotype with 500 or more representations in the WTCCC dataset and also having the largest *ORs* for disease-association of any haplotype meeting this specification. The reason for choosing only three regions was that, when more regions were added, there were an insufficient number observations to estimate the *ORs* for any of the possible

higher order combinations. These three regions were located at chromosomal locations 3p24.2, 14q24.1, and 16p13.13 and in the vicinity, respectively, of the genes EOMES, ZFP36L1, CLEC16A [13,14]. Chromosome 3; Region 22 (*d1*) spanned 0.65 mb of DNA and the 11-SNP-haplotype (number 234) was used [24]. Chromosome 14; Region 78 (*d2*) spanned 0.68 mb of DNA and the 3-SNP-haplotype (number 734) was used [24]. Chromosome 16; Region 85 (*d3*) spanned 0.20 mb of DNA and the SNP-haplotypes (numbers 814, 818, and 822) were combined into a single 15-SNP haplotype [24]. This was done because each of these risk-haplotypes were adjacent to each other and because the individual risk SNP haplotypes were part of the same extended 15-SNP-haplotype.

Haplotype Frequencies and Association Testing. Disease association tests, as measured by ORs and confidence intervals (CIs), were calculated for each of the CEHs and each of the 3 non-MHC risk haplotypes either alone or in different combinations. The WTCCC data was considered in its entirety and not further stratified. MS-associated haplotypes were analyzed by grouping them into five categories of CEHs, which consisted of: 1) (H+)-carrying CEHs (i.e. those containing the *HLA-DRB1*15:01~HLA-DOB1*06:02~a1* haplotype, *S2 File and Table S2*; 2) other increased risk or "extended risk" (*ER*) CEHs (*c23*, *c27*, *c34*, *c46*, *c68*, *c81*, *c85*, *c96*, and c107) as shown in S2 File (Table S3); 3) decreased risk or "all protective" (AP) CEHs (c5, c15, c18, c24, c30, c32, c51, and c73) as shown in S2 File (Table S3); 4) all CEHs not in the (H+), (ER), or (AP) groups (0) CEHs; and 5) the (c1) CEH by itself. We also explored a "protective" group, which excluded the (c5) CEH. However, this analysis is not presented because the findings were the same as when the AP group was analyzed as a whole. In many circumstances, an individual's MHC genotype was specified by the haplotype combination that they possessed. For example, by this convention, an individual homozygous for (H+) would be characterized as having the (H+,H+) MHC genotype. By contrast, a heterozygous individual would be characterized as having the (H+,0), the (H+,ER), the (H+,c1), or the (H+,AP) MHC genotype In the principal analysis, all MS-associations were assessed compared to a reference group consisting of the (0,0) MHC genotype. Similarly, when the disease associations for those "nonrisk" CEHs carrying the HLA-DRB1*03:01~HLA-DQB1*02:01~a6 haplotype were assessed, other carriers of this haplotype were also excluded from the (0,0) reference group. For notational simplicity, when using the (AP, AP) MHC genotype as a reference, this genotype was referred to as (AP^*) .

Disease associations for the risk SNP-haplotypes on Chromosomes 3, 14, and 16, were assessed compared to a reference group consisting of the (0,0) *MHC* genotype, and excluded individuals carrying their risk-haplotypes at these chromosomal locations. We designate (collectively) all non-risk-haplotypes at each of these chromosomal locations as the (0) haplotype

at each locus.

Significance of the differences between two *ORs* in disease association for any haplotype or haplotype combination was determined by z-scores calculated for the differences in the natural logarithm of the *ORs* for any two haplotypes. As discussed earlier, pair-wise comparisons of *ORs* are independent of the reference group chosen. The *MHC* genotype (0,0) had the largest sample size of any and, therefore, in order to maximize the statistical power to detect differences, the *ORs* used for pair-wise comparisons within the *MHC* were estimated relative to a reference group consisting of the (0,0) genotype at both the *MHC* and also at the any non-MHC locus included in the comparison. As noted in the *Introduction*, such a method eliminates the common reference group disease-risk to yield an estimate of the pairwise *RR*. Within the WTCCC cohort, we used a principal components (*PC*) analysis excluding *MHC* SNPs (Eigensoft) to correct the observations in *Tables S2 & S3* (*S2 File*) for the possible effects of geographic heterogeneity [25]. These adjustments did not significantly alter any of the associations shown in (*S2 File; Tables S 2& S3*).

Evaluating Additive and Multiplicative Risk-models. The ORs for the MHC alleles (H+, *ER*, and θ were determined relative to the (AP^*) reference group, which was assigned a value of $(R_b=R_{AP}=1)$. These observed ORs were used to estimate the RRs associated with each set of MHC alleles and, in turn, these RRs were used to assess the appropriateness of the additive and multiplicative risk-models for the different allelic combinations at the MHC. Subsequently, using a reference group consisting of the (0,0) MHC genotype, we determined the ORs for susceptibility alleles in the three non-MHC susceptibility regions -(d1), (d2) and (d3). The (0,0) MHC genotype was chosen as the reference because there were too few representations of the (AP, AP)MHC genotype in the WTCCC dataset. Nevertheless, these observed ORs were mathematically converted into ORs relative to the (AP, AP) MHC genotype and these re-referenced ORs, together with the ORs actually observed for the different allelic combinations at the MHC, were used to estimate the RRs associated with each allelic combinations at these four genomic locations (the MHC plus the three non-MHC susceptibility regions). These estimated RRs were then used to assess the appropriateness of the additive and multiplicative risk-models for the different allelic combinations of these four susceptibility regions. In all cases, only ORs estimated from combinations with ≥ 15 representations in the WTCCC were considered.

Acknowledgements

None.

References

- 1. Gourraud PA, Harbo HF, Hauser SL, Baranzini SE. (2012) The genetics of multiple sclerosis: an up-to-date review. *Immunol Rev* 248:87–103.
- 2. Hofker MH, Fu J, Wijmenga C. (2014) The genome revolution and its role in understanding complex diseases. *Biochim Biophys Acta* 1842:1889-1895.
- 3. Goodin DS. The nature of genetic susceptibility to multiple sclerosis: Constraining the Possibilities. *BMC Neurology* 2016;16:56.
- 4. Goodin DS. The Genetic and Environmental Bases of Complex Human-Disease: Extending the Utility of Twin-Studies. *PLoS One* 2012;7(12): e47875.
- GAMES, the Transatlantic Multiple Sclerosis Genetics Cooperative. (2003) A metaanalysis of whole genome linkage screens in multiple sclerosis. *J Neuroimmunol* 2003;143:39–46.
- 6. de Bakker PIW, Yelensky R, Pe'er I, et al. Efficiency and power in genetic association studies. *Nat Genet* 2005;37:1217-1223.
- 7. Herrera BM, Cader MZ, Dyment DA, et al. Multiple sclerosis susceptibility and the X chromosome. *Mult Scler* 2007;13:856–8.
- The Wellcome Trust Case Control Consortium & The Australo-Anglo-American Spondylitis Consortium. Associations can of 14,500 nonsynonymous SNPs in four diseases identifies autoimmunity variants. *Nature Genet* 2007;39:1329–1337.
- 9. The ANZgene Consortium. Genome-wide association study identifies new multiple sclerosis susceptibility loci on chromosomes 12 and 20. *Nature Genet* 2009;41:824–828.
- Baranzini SE, Wang J, Gibson RA, et al. Genome-wide association analysis of susceptibility and clinical phenotype in multiple sclerosis. *Hum Mol Genet*. 2009;18:767-778.
- De Jager PL, Jia X, Wang J, et al. Meta-analysis of genome scans and replication identify CD6, IRF8 and TNFRSF1A as new multiple sclerosis susceptibility loci. *Nature Genet* 2009;41:776–782.
- Sanna, S. Pitzalis M, Zoledziewska M, et al. Variants within the immunoregulatory CBLB gene are associated with multiple sclerosis. *Nature Genet* 2010;42:495–497.
- The International Multiple Sclerosis Genetics Consortium & the Wellcome Trust Case Control Consortium. Genetic risk and a primary role for cell-mediated immune

mechanisms in multiple sclerosis. Nature 2011;476:214-219.

- International Multiple Sclerosis Genetics Consortium (IMSGC). Analysis of immunerelated loci identifies 48 new susceptibility variants for multiple sclerosis *Nat Genet* 2014;45:1353-60.
- 15. Dyment DA, Herrera BM, Cader Z, et al. Complex interactions among MHC haplotypes in multiple sclerosis: susceptibility and resistance. *Hum Mol Genet* 2005;14:2019-2026.
- Hafler, DA, Compston A, Sawcer S, et al. Risk alleles for multiple sclerosis identified by a genomewide study. *N. Engl. J. Med.* 2007;357, 851–862.
- Ramagopalan, SV, Anderson, C, Sadovnick, AD, Ebers, GC. Genomewide study of multiple sclerosis. *N. Engl. J. Med.* 2007;357, 2199–2200.
- Link J, Kockum I, Lorentzen AR, et al. Importance of Human Leukocyte Antigen (HLA) Class I and II Alleles on the Risk of Multiple Sclerosis. *PLoS One* 2012 ;7(5):e36779.
- Patsopoulos NA, Barcellos LF, Hintzen RQ, et al. (2014) Fine-Mapping the Genetic Association of the Major Histocompatibility Complex in Multiple Sclerosis: HLA and Non-HLA Effects. *PLoS Genet* 9(11):e1003926.
- Chao MJ, Barnardo MC, Lincoln MR, Ramagopalan SV, et al. HLA class I alleles tag HLA-DRB1*1501 haplotypes for differential risk in multiple sclerosis susceptibility. *Proc Natl Acad Sci USA* 2008;105:13069-74.
- Lincoln MR, Ramagopalan SV, Chao MJ, Herrera BM, et al. Epistasis among HLA-DRB1, HLA-DQA1, and HLA-DQB1 loci determines multiple sclerosis susceptibility. *Proc Natl Acad Sci USA* 2009;106:7542-7.
- 22. Multiple Sclerosis Genetics Group. Linkage of the MHC to familial multiple sclerosis suggests genetic heterogeneity. *Hum Molec Genet* 1998;7:1229–1234.
- Goodin DS, Khankhanian P. Single Nucleotide Polymorphism (SNP)-Strings: An Alternative Method for Assessing Genetic Associations. *PLoS One* 2014;9(4):e90034.
- Khankhanian P, Gourraud PA, Lizee A, Goodin DS. Haplotype-based approach to known MS-associated regions increases the amount of explained risk. *J Med Genet*. 2015;52:587-594.
- Gragert L, Madbouly A, Freeman J, Maiers M. Six-locus high resolution HLA haplotype frequencies derived from mixed-resolution DNA typing for the entire US donor registry. *Hum Immunol* 2013;74:1313-1320.

- Goodin DS, Khankhanian P, Gourraud PA, Vince N. (2018) Highly conserved extended haplotypes of the major histocompatibility complex and their relationship to multiple sclerosis susceptibility. *PLoS One* 13(2):e0190043.
- Liguori M, Marrosu MG, Pugliatti M. et al. Age at onset in multiple sclerosis. *Neurol Sci* 2000;21:S825-S829.
- Grytten TN, Lie SA, Aarseth JH, Nyland H, Myhr KM (2008) Survival and cause of death in multiple sclerosis: results from a 50-year follow-up in Western Norway. *Mult Scler* 2008;14: 1191–1198.
- 29. Ragonese P, Aridon P, Mazzola MA, Callari G, Palmeri B, et al. Multiple sclerosis survival: a population-based study in Sicily. *Eur J Neurol* 2010;17: 391–397
- Kingwell E, van der KM, Zhao Y, Shirani A, Zhu F, et al Relative mortality and survival in multiple sclerosis: findings from British Columbia, Canada. *J Neurol Neurosurg Psychiatry* 2012;83: 61–66.
- 31. Rosati G. The prevalence of multiple sclerosis in the world: an update. *Neurol Sci.* 2001;22:117–39.
- Sundström P, Nyström L, Forsgren L. Incidence (1988–97) and prevalence (1997) of multiple sclerosis in Västerbotten County in northern Sweden. *J Neurol Neurosurg Psychiatry*. 2003;74:29–32.
- 33, Harding K, Zhu F, Alotaibi MD, Dugan T, Tremlett H, Kingwell E. Causes that contribute to deaths due to multiple sclerosis: analysis of population-based multiplecause-death data. Presentation 144. ECTRIMS 2018, Berlin.
- Vost A, Wolochow D, Howell D. Incidence of infarcts of the brain in heart diseases. J Path Bact 1964;88:463-470.
- Georgi VW. Multiple Sklerose: Pathologisch-Anatomische Befunde multiple Sklerose bei klinisch nicht diagniostizierte Krankbeiten. *Schweiz Med Wochenschr* 1966;20:605-607.
- 36. Gilbert J, Sadler M. Unsuspected multiple sclerosis. Arch Neurol 1983;40:533-536.
- Engell T. A clinical patho-anatomical study of clinically silent multiple sclerosis. *Acta Neurol Scand* 1989;79:428-430.

- Okuda DT, Mowery EM, Cree BAC, Crabtree EC, Goodin DS, Waubant E, Pelletier D. Asymptomatic spinal cord lesions predict disease progression in radiologically isolated syndrome. *Neurology* 2011;76:686-692.
- Witte JS, Carlin JB, Hopper JL. Likelihood-Based Approach to Estimating Twin concordance for dichotomous traits. *Genetic Epidemiol*. 1999;16:290–304.
- Willer CJ, Dyment DA, Rusch NJ, Sadovnick AD, Ebers GC, the Canadian Collaborative Study Group. Twin concordance and sibling recurrence rates in multiple sclerosis. *Proc Natl Acad Sci U S A*. 2003;100:12877–82.
- 41. French Research Group on Multiple Sclerosis. Multiple sclerosis in *54* twinships: Concordance rate is independent of zygosity. *Ann Neurol* 1992;32:724-727.
- 42. Mumford CJ, Wood NW, Kellar-Wood H, Thorpe JW, Miller DH, Compston DA. The British Isles survey of multiple sclerosis in twins. *Neurology*. 1994;44:11–5.
- 43. Hansen T, Skytthe A, Stenager E, et al. Concordance for multiple sclerosis in Danish twins: An update of a nationwide study. Mult Scler 2005;11:504–510.
- 44. Islam T, Gauderman WJ, Cozen W, et al. Differential twin concordance for multiple sclerosis by latitude of birthplace. *Ann Neurol* 2006; 60: 56–64.
- 45. Ristori G, Cannoni S, Stazi MA, et al. and the Italian Study Group on Multiple Sclerosis in Twins. Multiple sclerosis in twins from continental Italy and Sardinia: A Nationwide Study Ann Neurol 2006;59:27–34.
- Kuusisto H, Kaprio J, Kinnunen E, et al. Concordance and heritability of multiple sclerosis in Finland: Study on a nationwide series of twins. *Eur J Neurol* 2008;15: 1106–1110.
- Hansen T, Skytthe A, Stenager E, Petersen HC, Brønnum-Hansen H, Kyvik KO.
 Concordance for multiple sclerosis in Danish twins: an update of a nationwide study.
 Mult Scler. 2005;11:504–10.
- Hansen T, Skytthe A, Stenager E, Petersen HC, Kyvik KO, Brønnum-Hansen H. Risk for multiple sclerosis in dizygotic and monozygotic twins. *Mult Scler*. 2005;11:500–3.

- 49. O'Gorman C, Lin R, Stankovich J, Broadley SA. Modeling genetic susceptibility to multiple sclerosis with Family Data. *Neuroepidemiology* 2013;40:1-12.
- 50. Goodin DS. The causal cascade to multiple sclerosis: A model for MS pathogenesis. *PLoS One* 2009;4(2):e4565.
- 51. Goodin DS. The epidemiology of multiple sclerosis: Insights to a causal cascade. *Handb Clin Neurol.* 2016;138:173-206.
- 52. Jacobson HI. The maximum variance of restricted unimodal distributions. *Ann Math Stat.* 1969;40:1746–52.
- 53. Freeman JB, Dale R. Assessing bimodality to detect the presence of a dual cognitive process. *Behav Res.* 2013;45:83–97.
- 54. Goodin The genetic basis of multiple sclerosis: a model for MS susceptibility. *BMC Neurology* 2010, 10:101.
- 55. Siemiatycki J, Thomas DC. Biological models and statistical interactions: An example from multistage carcinogenesis. *Int J Epidemiol*. 1981;10:383-387.
- 56. Kodell RL, Gaylor DW. On the additive and multiplicative models of relative risk. *Biometrical J* 1989;31:359-370.
- 57. Greenland S. Additive Risk versus Additive Relative Risk Models. *Epidemiology* 1993;4:32-36.
- Rothman KJ, Greenland S. Modern Epidemiology. Lippincott, Williams & Wilkins, Philadelphia. PA, 1998
- van der Mei I, Lucas R, Taylor B, et al. Population attributable fractions and joint effects of key risk factors for multiple sclerosis. *Mult Scler J*. 2016;22:461–469.
- 60. De Jager PL, Chibnik LB, Cui J, et al. Integrating genetic risk factors into a clinical algorithm for multiple sclerosis susceptibility. *Lancet Neurol* 2009;8:1111-1119.
- 61. Gourraud PA, McElroy JP, Caillier SJ, et al. Aggregation of MS genetic risk variants in multiple and single case families. *Ann Neurol* 2011;69:65-74.
- 62. Isobe N, Damotte V, Lo Re M, et al. Genetic Burden in multiple sclerosis families. *Genes and Immunity* 2013;14:434-440.
- 63. Viera AJ. Odds ratios and risk ratios: What's the difference and why does it matter?.

South Med J. 2008;101:730-734.

- Xie T, Rowen L, Aquado B, et al. Analysis of the Gene-Dense Major Histocompatibility Complex Class III Region and Its Comparison to Mouse. *Genome Res* 2003;134:2621-2636.
- 65. Ascherio, A., Munger, K.L. Environmental risk factors for multiple sclerosis. Part I: the role of infection. *Ann. Neurol.* 2007;61, 288–299.
- Ascherio, A., Munger, K.L.. Environmental risk factors for multiple sclerosis. Part II: noninfectious factors. *Ann. Neurol.* 2007;61, 504–513.
- 67. Ascherio, A., Munger, K.L., Simon, K.C., 2010. Vitamin D and multiple sclerosis. *Lancet Neurol.* 2010;9, 599–612.
- 68. Orton SM, Herrera BM, Yee IM, et al., and the Canadian Collaborative Study Group.
 (2006) Sex ratio of multiple sclerosis in Canada: A longitudinal study. *Lancet Neurol*. 5:932–6.
- Bager P, Nielsen NM, Bihrmann K, Frisch M, Wohlfart J, et al. (2006) Sibship characteristics and risk of multiple sclerosis: A nationwide cohort study in Denmark. *Am J Epidemiol* 163:1112–1117.
- 70. Compston A, Coles A (2002) Multiple sclerosis. *Lancet* 359:1221–31.
- Dyment DA, Yee IML, Ebers GC, Sadovnick AD, and the Canadian Collaborative Study Group (2006) Multiple sclerosis in step siblings: Recurrence risk and ascertainment. J Neurol Neurosurg Psychiatry 77:258–259.
- 72. Ebers GC, Sadovnick AD, Dyment DA, Yee IM, Willer CJ, et al. (2004) Parent-of-origin effect in multiple sclerosis: observations in half-siblings. *Lancet* 363:1773–1774.
- 73. Ebers GC, Yee IML, Sadovnick AD, Duquette P, and the Canadian Collaborative Study Group (2000) Conjugal multiple sclerosis: Population based prevalence and recurrence risks in offspring. *Ann Neurol* 48:927–931.
- Sadovnick AD, Yee IML, Ebers GC, and the Canadian Collaborative Study Group (2005)
 Multiple sclerosis and birth order: A longitudinal cohort study. *Lancet Neurol* 4:611–617.
- Sadovnick AD, Ebers GC, Dyment DA, Risch NJ, and the Canadian Collaborative Study Group (1996) Evidence for genetic basis of multiple sclerosis. *Lancet* 347:1728–1730.

- Schork NJ, Murray SS, Frazer KA, Topol EJ. Common vs. rare allele hypotheses for complex diseases. *Curr Opin Genet Dev.* 2009;19: 212–219.
- 77. Roux T, Courtillot C, Debs R, et. al.. Fecundity in women with multiple sclerosis: an observational mono-centric study. *J Neurol*. 2015;262:957–960.
- 78. Zuk O, Hechter E, Sunyaev SR, Lander ES. The mystery of missing heritability: Genetic interactions create phantom heritability. *Proc Natl Acad Sci U S A*. 2012;109:1193-8
- Lill CM. Recent advances and future challenges in the genetics of multiple sclerosis. Front Neurol 2014;5:130
- Bashinskaya VV, Kulakova OG, Boyko AN, Favorov AV, Favorova OO A review of genome-wide association studies for multiple sclerosis: classical and hypothesis-driven approaches. *Hum Genet*. 2015;134:1143-62.
- 81. Poser, C. M. et al. New diagnostic criteria for multiple sclerosis: guidelines for research protocols. *Ann Neurol* 13, 227-231 (1983).
- McDonald, W. I. et al. Recommended diagnostic criteria for multiple sclerosis: guidelines from the International Panel on the diagnosis of multiple sclerosis. *Ann Neurol* 50, 121-127 (2001).
- Polman, C. H. et al. Diagnostic criteria for multiple sclerosis: 2005 revisions to the "McDonald Criteria". *Ann Neurol* 58, 840-846 (2005).
- 84. Zheng X, J Shen J, Cox C, et al. (2014) HIBAG HLA genotype imputation with attribute bagging. *Pharmacogenom J* 14:192–200.
- 85. Gourraud PA, Lamiraux P, El-Kadhi N, et al. (2005) Inferred HLA haplotype information for donors from hematopoietic stem cells donor registries. *Hum Immunol* 66:563–70.
- Gourraud PA, Khankhanian P, Cereb N, et al. (2014) HLA diversity in the 1000 genomes dataset. *PLoS One* 9:e9782.

Supporting Information Captions

File S1. This *Supplemental File* describes, in detail, the susceptibility Model used for this manuscript and rigorously develops its logical framework. It includes the methods used to estimate both the "population" parameters such as:

 $P(MS), P(MS | MZ_{MS}), P(H+), P(H-), P(F), P(MS | F), P(MS | M), etc.$

as well as the "non-population" parameters of interest such as:

 $P(MS | IG_{MS}), P(G), P(MS | G), P(MS | G, F), P(MS | G, H+), P(F | G), etc.$

(See Table 1, Main Text for the definitions of the different parameters used in the Model)

It also describes, in detail, the manner by which the values of these non-population parameters can be estimated from the directly observable population parameters.

File S2. This *Supplemental File* describes composition of the *CEHs* found in the WTCCC dataset as well as their individual relationships to MS susceptibility and also how this *CEH* composition differs between populations around the world.

Furthermore, this file considers the theoretical underpinnings for the commonly used additive and multiplicative Models for the accumulation of disease "risk" with increasing number of "risk haplotypes" being present in an individual's genotype

Table 1. Definitions for Epidemiological Parameters used in the Model*

(Z)	Set of all individuals in the population
P(MS)	Life-time probability of getting <i>MS</i> for members of (<i>Z</i>)
(G)	Set of all individuals in (<i>Z</i>) who have some chance of getting MS
(G-)	Set of all individuals in (Z) who have no chance of getting MS
(H+)	Set of all carriers of the <i>DRB1*15:01~DQB1*06:02~a1</i> haplotype in (<i>Z</i>)
(H-)	Set of all non-carriers of the $(H+)$ HLA haplotype in (Z)
(<i>M</i>)	Set of all men in (Z)
(F)	Set of all women in (Z)
<i>P</i> (<i>E</i>)	Probability of an environmental exposure "sufficient to cause MS" in the set (G) , given the prevailing environmental conditions of the time
$P(MS \mid MZ_{MS})$	Life-time probability of getting <i>MS</i> in an <i>MZ</i> -twin whose co-twin either has, or will develop, <i>MS</i>
$P(MS \mid IG_{MS})$	Value of $P(MS \mid MZ_{MS})$, which has been adjusted to exclude the impact of the similar intrauterine and early post-natal environments of MZ -twins

Parameter Definition

* A complete presentation of the Model and more precise definitions of these terms are

provided in File S1.

 Table 2. Canadian Population Data on MZ-Twin Concordance broken down by

(H+)-haplotype and Gender-Status *

MZ-Twins

HLA-DRB1*15 Status	H+	H–	Totals
Concordant for MS (C)	9	11	20
Discordant for MS (D)	31	42	73
Totals	40	53	93
Pair-wise Concordance	0.23	0.21	0.22
Proband-wise Concordance	0.31	0.29	0.30
Gender Status	Women	Men	Totals
<i>Gender Status</i> Concordant for MS (C)	Women 22	<i>Men</i> 2	Totals 24
<i>Gender Status</i> Concordant for MS (C) Discordant for MS (D)	<i>Women</i> 22 66	<i>Men</i> 2 43	Totals 24 109
<i>Gender Status</i> Concordant for MS (C) Discordant for MS (D) Totals	<i>Women</i> 22 66 88	<i>Men</i> 2 43 45	<i>Totals</i> 24 109 133
<i>Gender Status</i> Concordant for MS (C) Discordant for MS (D) Totals Pair-wise Concordance	<i>Women</i> 22 66 88 0.25	<i>Men</i> 2 43 45 0.04	<i>Totals</i> 24 109 133 0.18

Other Canadian Data:

P(H+) = 0.24 $P(F) \approx 0.5$

 $P(H+|MZ_{MS}) = 40/93 = 0.43$ $P(H+|MS,MZ_{MS}) = 9/20 = 0.45$ $P(H+|MZ_{MS}) = 1.79 * P(H+)$ $P(H+|MS,MZ_{MS}) = 1.04 * P(H+|MZ_{MS})$

 $P(F | MZ_{MS}) = 88 / 133 = 0.66$ $P(F | MS, MZ_{MS}) = 22 / 24 = 0.92$ $P(F | MZ_{MS}) = 1.32 * P(F)$ $P(F | MS, MZ_{MS}) = 1.39 * P(F | MZ_{MS})$

Time Period (#1) of 1941-1945: $P(F | MS)_1 / P(M | MS)_1 = 2.2$ Time Period (#2) of 1976-1980: $P(F | MS)_2 / P(M | MS)_2 = 3.2$

* Data from Willer et al. [40] and Orton et al. [68]

the *MZ*-twins were drawn from the 19,938 MS-patients in the CCGPSMS database
 Pair-wise concordance calculated as: C/(C+D)
 Proband-wise concordance calculated as: 2C/(2C+D)
 adjusted [39] for double ascertainments (13/24=54%)

Table 3. Impact of *CEHs* Frequency on Disease Association

MHC Haplotype

Combinations	Odds Ratio [*]	p-value**
(H+)-haplotype (<50) † -1 copy	3.2 (3.0 – 3.6)	< E-148
(H+)-haplotype (<50) \dagger -2 copies	4.3 (2.8 – 6.5)	< E-12
(H+)-haplotype (\geq 50) † -1 copy	2.9 (2.8 - 3.1)	< E-277
(H+)-haplotype (\geq 50) † -2 copies	6.6 (5.5 - 7.9)	<e-117< td=""></e-117<>
(H+)-haplotype (All) † -1 copy	3.0 (2.8 - 3.2)	< E-300
(H+)-haplotype All) † -2 copies	6.4 (5.6 – 7.3)	< E-200
(A2)-haplotype (<50) † -1 copy	1.7 (1.3 – 2.1)	<e-4< td=""></e-4<>
(A2)-haplotype (<50) † -2 copies	na	na
(A2)-haplotype (\geq 50) [†] -1 copy	1.7 (1.4 – 2.0)	< E-8
(A2)-haplotype (\geq 50) [†] -2 copies	1.9 (0.4 - 8.3)	ns
(A2)-haplotype (All) † -1 copy	1.7 (1.5 – 1.9)	<e-12< td=""></e-12<>
(A2)-haplotype (All) [†] 2 copies	2.9 (1.4 - 6.0)	< E-2
(H+)-(A2) genotype	4.5 (3.4 – 6.1)	< E-28
(A6)-haplotype (<50) † -1 copy	1.0 (0.8 – 1.2)	ns
(A6)-haplotype (<50) † -2 copies	0.7 (0.2 – 2.7)	ns
(A6)-haplotype (\geq 50) [†] -1 copy	1.0 (0.9 – 1.1)	ns
(A6)-haplotype (\geq 50) [†] -2 copies	2.9 (2.2 - 3.8)	<e-15< td=""></e-15<>
non-c1 - (A6)-haplotype (\geq 50) † – 1 copy	0.9 (0.7 - 1.1)	ns
non-c1 - (A6)-haplotype (\geq 50) † – 2 copies	1.7 (0.3 – 10.1)	ns
(A14)-haplotype (<50) † -1 copy	2.2 (1.8 – 2.8)	<e-13< td=""></e-13<>
(A14)-haplotype (<50) † -2 copies	na	na
(A14)-haplotype (\geq 50) [†] -1 copy	2.0 (1.5 - 2.6)	< E-6
(A14)-haplotype (\geq 50) –2 copies	na	na
(A14)-haplotype (All) † –1 copy	2.1 (1.8 – 2.5)	<e-17< td=""></e-17<>
(A14)-haplotype (All) † -2 copies	na	na
(H+)-(A14) genotype	5.7 (4.0 - 8.1)	< E-25
(A14)-(A2) genotype	5.6 (2.0 - 16.3)	< E-3

* Odds ratio (OR) of disease for individuals having particular haplotype combinations compared to the (0,0) MHC genotype. In the case of *the A6 haplotypes*, the comparison group also did not possess of these haplotypes either.

- ** The p-values are expressed in scientific notation as powers of 10 (E); (na=not available; ns=not significant)
- † (H+)-haplotype = HLA-DRB1*15:01~HLA-DQB1*06:02~a1
 (A2)-haplotype = HLA-DRB1*03:01~HLA-DQB1*02:01~a2
 (A6)-haplotype = HLA-DRB1*03:01~HLA-DQB1*02:01~a6
 (A14)-haplotype = HLA-DRB1*13:03~HLA-DQB1*03:01~a14
 - (<50) = CEHs with fewer than 50 representations in the WTCCC (≥ 50) = CEHs with 50 or more representations in the WTCCC

Table 4. Impact on Phenotype of Combining MHC CEHs into a Genotype^{\dagger}

MHC Haplotype		
Combinations	Odds Ratio [*]	p-value**
c2 –1 copy	3.2 (2.9–3.5)	< E-153
c2 –2 copies	5.7 (3.4–9.7)	< E-12
c3 –1 copy	2.2 (2.0-2.5)	< E-36
c3 –2 copies	2.6 (1.3-5.4)	< E-2
c2 + other (H+)-haplotype ^{††}	7.5 (6.0–9.3)	< E-98
$c3 + other (H+)-haplotype^{\dagger\dagger}$	6.1 (4.6-8.0)	< E-48
c5 –1 copy	0.5 (0.4–0.6)	< E-11
$c^{2} + c^{5}$	1.5 (0.7–2.9)	ns
<i>c</i> 3 + <i>c</i> 5	0.3 (0.1–1.4)	ns
Protective (non-c5) ^{††} –1 copy	0.5 (0.4–0.6)	< E-14
<i>Protective (non-c5)</i> ^{\dagger+} -2 copies	0.2 (0.1–0.7)	< E-2
All Protective ^{††} –1 copy	0.5 (0.4–0.6)	< E-23
All Protective ^{\dagger†-2 copies}	0.13 (0.04–0.4)	< E-4
$c2 + All Protective^{\dagger \dagger}$	2.1 (1.4–3.1)	< E-3
$c3 + All Protective^{\dagger \dagger}$	1.0 (0.6–1.8)	ns
(H+)-haplotype ^{\dagger†} –1 copy	3.0 (2.8–3.2)	< E-300
(H+)-haplotype ^{\dagger†} -2 copies	6.4 (5.6–7.3)	< E-200
(H+)-haplotype + All Protective ^{\dagger†}	1.8 (1.4–2.2)	< E-7
(H+)-haplotype $^{\dagger\dagger}+c5$	1.5 (1.1–2.0)	< 0.05
Extended Risk ^{††} –1 copy	2.0 (1.7–2.3)	< E-21
Extended $Risk^{\dagger\dagger} - 2$ copies	5.0 (1.5–16.5)	< E-2
All Protective + Extended Risk ^{$\dagger\dagger$}	1.0 (0.4–2.3)	ns
(H+)-haplotype + Extended Risk ^{\dagger†}	4.3 (3.3–5.7)	< E-29
c1–1 copy	1.1(1.0–1.2)	ns
c1–2 copies	3.2 (2.4–4.4)	< E-13
c1 + (H+)-haplotype ^{††}	4.0 (3.5–4.7)	< E-86
$c1 + All Protective^{\dagger\dagger}$	0.4 (0.3–0.7)	< E-3
$c1$ + Extended Risk ^{$\dagger \dagger$}	3.8 (2.3-6.1)	< E-7

† Haplotype names (*e.g.*, *c1*, *c2*, *etc*) are defined in the *S2 File* (*Tables S2 &S3*).

 \dagger \dagger (H+)-haplotype = HLA-DRB1*15:01~HLA-DQB1*06:02~a1

Protective (non-c5) = (c15, c18, c24, c30, c32, c51, and c73) All Protective (AP) = (c5, c15, c18, c24, c30, c32, c51, and c73) Extended Risk (ER) = (c23, c27, c34, c46, c68, c81, c85, c96, and c107)

Risk combinations were defined as:

"single copy risk" = 1 copy of any (H+)-CEH or any (ER) CEH

and:

"double copy risk" = 2 copies of any (H+)-CEHs, (c1), or 2 copies of any (ER)CEH or the combinations of (H+ and ER), (H+ and c1), and (ER and c1).

- * Odds ratio (OR) of disease for individuals having certain haplotype combinations compared to having the (0,0) MHC genotype. The 95% confidence intervals (*CIs*) are shown in parentheses.
- ** The p-values are expressed in scientific notation as powers of 10 (E); ns=not significant.

Table 5. Impact on Phenotype of Combining Non-MHC Genotypes[†]

Combinations [†]	Odds Ratio*	p-value**
d1 –1 copy	1.9 (1.6–2.3)	< E-12
d1–2 copies	2.6 (1.6-4.2)	< E-4
d2 –1 copy	1.1 (1.0–1.2)	< 0.05
d2–2 copies	1.2 (1.1–1.4)	< E-2
d3–l copy	1.2 (1.1–1.2)	< E-3
d3–2 copies	1.5 (1.3–1.6)	< E-13
$1 \operatorname{copy} d1 + 1 \operatorname{copy} d2$	1.9 (1.4–2.5)	< E-5
$1 \operatorname{copy} d1 + 2 \operatorname{copies} d2$	3.3 (1.6–7.1)	< E-4
2 copies d1 + 1 copy d2	4.2 (2.1-8.7)	< E-3
$1 \operatorname{copy} d1 + 1 \operatorname{copy} d3$	2.5 (2.0-3.3)	< E-12
$1 \operatorname{copy} d1 + 2 \operatorname{copies} d3$	2.0 (1.3-3.1)	< E-2
2 copies d1 + 1 copy d3	2.5 (1.2–5.5)	< 0.05
$1 \operatorname{copy} d2 + 1 \operatorname{copy} d3$	1.2 (1.1–1.4)	< E-4
1 copy d2 + 2 copies d3	1.4(1.1–1.7)	< E-2
2 copies d2 + 1 copy d3	1.5 (1.3–1.7)	< E-6
2 copies $d2 + 2$ copies $d3$	2.4 (1.7–3.4)	< E-6
$1 \operatorname{copy} d1 + 1 \operatorname{copy} d2 + 1 \operatorname{copy} d3$	2.6 (1.8-4.0)	< E-5
$1 \operatorname{copy} d1 + 1 \operatorname{copy} d2 + 2 \operatorname{copies} d3$	1.5 (0.7–3.1)	ns

Non-MHC Genotype

† Risk haplotypes [24] for Non-MHC susceptibility loci (see text)

d1 = Region 22 in Chromosome 3
d2 = Region 78 in Chromosome 14
d3 = Region 85 in Chromosome 16

* Odds ratio (OR) of disease for individuals having certain haplotype combinations compared to having the (0,0) genotype at both the MHC and the non-MHC loci.

** The p-values are expressed in scientific notation as powers of 10 (E); ns=not significant.

Figure Legends

Figure 1. Lower triangular plot of the natural logarithm of odds ratios (ORs) and z-scores for the difference in disease risk for different two-MHC genotype combinations. To maximize statistical power, prior to calculating the comparative values for ln(OR), all ORs and standard deviations for each genotype were estimated relative to the MHC genotype (0,0) and then the combination compared to each other both as a ratio and as a z-score – see *Methods* and *Introduction*. Only genotypes with combinations of haplotypes with (H+), "extended risk" (ER), "all protective (AP), (c1), and (0) are shown. These genotypes are listed on both the x-axis (as columns) and y-axis (as rows) and the ORs and z-scores for each two-genotype comparison are represented as numbers at the points of intersection of the column and row for any two genotypes. Comparisons with an absolute z-score < 2.0, are shaded in yellow; comparisons with an absolute z-score = 2.0-3.0 are shaded in pale blue or pale red; comparisons with an absolute z-score = 3.1-7.0 are shaded in blue or red; comparisons with an absolute z-score > 7.0 are shaded in dark blue or dark red. Positive numbers (red shades) indicate that the genotype in the column has a greater OR than the genotype in the row. Conversely, negative numbers (blue shades) indicate that the genotype in the column has a smaller OR than the genotype in the row. For example, the genotype (AP-2) has a smaller OR than the genotype (AP-1). Similarly, the genotype (H-1) has a smaller OR than the combination (H-2). ORs given as (0.0) indicate that (OR<0.05). For no genotype were there zero MS cases observed. The following designations are use to indicate the different genotype configurations:

AP1	=	1 copy of "All Protective" (AP) + 1 copy of (0)
AP2	=	2 copies of "All Protective" (AP)
H-1	=	1 copy of (H+) + 1 copy of (0)
H-2	=	2 copies of (H+)
ER-1	=	1 copy of "Extended Risk" (ER) + 1 copy of (0)
ER-2	=	2 copies of "Extended Risk" (ER)
c1-1	=	1 copy of the c1 CEH + 1 copy of (0)
c1-2	=	2 copies of the c1 CEH
AP-ER	=	1 copy of "AP" + 1 copy of "ER"
AP - H	=	<i>l copy of "AP" + 1 copy of (H+)</i>
AP-c1	=	<i>l copy of "AP" + 1 copy of the c1 CEH</i>
ER-c1	=	1 copy of "ER" + 1 copy of the c1 CEH
H-c1	=	1 copy of (H+) + 1 copy of the c1 CEH
H-ER	=	1 copy of (H+) + 1 copy of "ER
0,0	=	2 copies of (0)

Figure 2. Lower triangular plots of the natural logarithm of the odds ratios (*ORs*) and z-scores for the different transitions from one MHC haplotype to another in different genotypic contexts. For example, the point of intersection for $(0 \rightarrow H^+)$ and $(c1 \rightarrow H^+)$ represents the ratio of:

OR for the transition:
$$(c1,0) \rightarrow (c1,H+)$$

OR for the transition: $(H+,0) \rightarrow (H+,H+)$

or equivalently:

 $\frac{OR \text{ for the transition: } (c1,0) \rightarrow (H+,0)}{OR \text{ for the transition: } (c1,H+) \rightarrow (H+,H+)}$

Only transitions for (H^+) , "extended risk" (ER), "all protective (AP), (c1), and (0) haplotypes are shown. These transitions are indicated on both the x-axis (as columns) and y-axis (as rows) and values for ln(OR) and the z-scores for each transition comparison are represented as numbers at the points of intersection of the column and row for any two genotypes. The designation "na" indicates data "not available". Comparisons with an absolute z-score >3.0, are shaded either dark blue (negative) or dark red (positive); comparisons with an absolute z-score = 2.0-3.0 are shaded either light blue (negative) or light red (positive); comparisons with an absolute z-score = 1.0-2.0are shaded either pale blue (negative) or pale red (positive) yellow; comparisons with absolute zscores <1.0 are shaded in vellow: Positive numbers indicate that OR for the transition (indicated by the column) is greater for the 1st genotypic configuration (indicated by the row) than it is for the 2^{nd} . Conversely, a negative number indicates that the transition *OR* for the 2^{nd} genotypic configuration is greater than the 1st. For example, the number (3.4) in the 1st column, 5th row of the z-score table, indicates that the OR for transition from genotype (AP, 0) to (AP, H^+) is significantly greater that the OR for the transition from $(H+, \theta)$ to (H+, H+). Similarly, the number (-2.2) in the 1st column, 3rd row of the z-score table, indicates that the OR for transition from genotype (0,0) to $(0,H^+)$ is significantly less that the OR for the transition from (c1,0) to $(c1,H^+)$. Because of symmetry, the OR for comparing the transition from genotype (AP, 0) to (AP, H^+) with the transition from genotype (H+,0) to (H+,H+) is mathematically equivalent to the OR for comparing the transition from genotype (AP, 0) to $(H^+, 0)$ with the transition from genotype (AP,H+) to (H+,H+). Therefore, the interpretation for the meaning of the rows and columns is completely interchangeable (although the implication of positive and negative numbers remains unchanged).

Figure 3. Natural logarithm of the odds ratios (*ORs*) and z-scores for the different combinations of *MHC* and non-*MHC* genotypes. All *ORs* were calculated relative to a group consisting of the same *MHC* genotype and with the genotypes (0,0) at all non-MHC loci involved in the comparison – see text. The *MHC* genotypes, in order of increasing disease-risk (Table 4), are presented on the *x*-axis (as columns) and the genotypes at non-*MHC* loci are presented on the *y*-axis (as rows). The values for *ln(OR)* and the *z*-scores for each comparison are represented as numbers at the points of intersection of the column and row for any two haplotypes. Comparisons with a z-score (|z| < 1) are shaded in yellow; comparisons with a z-score ($2 \ge z \le 3$) are shaded in light red; comparisons with a z-score ($3 \ge z \le 4$) are shaded in red; comparisons with a z-score ($z \ge 4$) are maroon. Specific combinations having marginal totals of less than 15 representations in the WTCCC are indicated by (*na*). Of the 83 observations presented, only 12% had a marginal total of less than 25 and 13% had a marginal total from 25 to less than 50.

Figure 4. Conformity of the observed effect of combining different MHC haplotypes with an additive and a multiplicative model of combined risk. Yellow bands represent the definitional odds ratios (*ORs*) relative to a reference group consisting of the (*AP,AP*) or (*AP**) genotype (i.e., as defined in the text: $R_b = R_{AP*} = 1$). With the exception of (*c1*), which seems to behave in a unusual fashion, the combination of other risk alleles produced, in general, a risk in between the two models, albeit closer to that predicted by the additive model. All combinations had, at least, 50 representations in the WTCCC and the green shading indicates the *ORs* actually observed. Cells with yellow shading in the "Observed" column also represents the *ORs* actually observed. However, in these yellow-highlighted cases, the *ORs* were used to approximate the relative risks (*RRs*), which, in turn, were used to assess whether the genotypes that are not yellow-highlighted conformed to the additive and multiplicative models (*see Methods*).

Figure 5. Conformity of the observed effect of combining different genotypes at the MHC and one susceptibility region with an additive and a multiplicative model of combined risk. The non-MHC susceptibility haplotypes are: (*d1*); (*d2*); and (*d3*) – see *Methods*. Yellow bands, as in Fig. 4, represent the definitional *ORs* for different non-MHC genotypes actually observed, but which have been re-referenced to a group with the (*AP,AP*) MHC genotype. The *ORs* for all MHC genotypes are also those actually observed (Fig 4). Only haplotype combinations with ≥ 15 or more representations in the WTCCC are shown. Combinations with fewer than 50 representations are shaded in pink; combinations with at least 50 representations are shaded in green.

Figure 6. Conformity of the observed effect of combining different genotypes at the MHC and two susceptibility regions with an additive and a multiplicative model of combined risk. The non-MHC susceptibility haplotypes are: (*d1*); (*d2*); and (*d3*) – see *Methods*. The *ORs* listed are those actually observed (Figs 4 & 5). Only haplotype combinations with \geq 15 or more representations in the WTCCC are shown. Combinations with fewer than 50 representations are shaded in pink; combinations with at least 50 representations are shaded in green.

Figure 7. Conformity of the observed effect of combining different genotypes at the MHC and three susceptibility regions with an additive and a multiplicative model of combined risk. The non-MHC susceptibility haplotypes are: (*d1*); (*d2*); and (*d3*) – see *Methods*. The *ORs* listed are those actually observed (Figs 4 & 5). Only haplotype combinations with \geq 15 or more representations in the WTCCC are shown. Combinations with fewer than 50 representations are shaded in pink; combinations with at least 50 representations are shaded in green.

In(OR)

	AP-1	AP-2	H-1	H-2	ER-1	ER-2	c1-1	c1-2	AP-H	AP-ER	AP-c1	ER-c1	H-c1	H-ER
AP-2	1.4													
H-1	-1.8	-3.1												
H-2	-2.5	-3.9	-0.7											
bioRxiv preprint (certified by peer re	loi: https://doi.c view) is the au	o <mark>rg/10.11017603</mark> hor/funder. Thi	878: this versions article is a US	n posted April Government v	10, 2019. The o ork. It is not su	opyright holder bject to copyrig	for this preprir ht under 17 US	t (which was no C 105 and is a)t so					
ER-2	-2.3	-3.6	-0.5	0.2	-0.9									
c1-1	-0.7	-2.1	1.0	1.76	0.6	1.5								
c1-2	-1.8	-3.2	-0.1	0.7	-0.5	0.4	-1.1							
AP-H	-1.2	-2.6	0.5	1.3	0.1	1.0	-0.5	0.6						
AP-ER	-0.6	-2.0	1.1	1.9	0.7	1.6	0.1	1.2	0.6					
AP-c1	0.2	-1.2	2.0	2.73	1.6	2.5	1.0	2.1	1.5	0.8				
ER-c1	-1.9	-3.3	-0.23	0.5	-0.7	0.3	-1.2	-0.2	-0.7	-1.3	-2.2			
H-c1	-2.0	-3.4	-0.3	0.5	-0.7	0.2	-1.3	-0.2	-0.8	-1.4	-2.3	0.1		
H-ER	-2.1	-3.5	-0.4	0.4	-0.8	0.1	-1.4	-0.3	-0.9	-1.5	-2.3	0.2	-0.1	
0,0	-0.7	-2.0	1.1	1.8	0.7	1.6	0.09	1.2	0.6	0.0	-0.9	1.3	1.4	1.5

z-scores

	AP-1	AP-2	H-1	H-2	ER-1	ER-2	c1-1	c1-2	AP-H	AP-ER	AP-c1	ER-c1	H-c1	H-ER
AP-2	2.3													
H-1	-24.9	-5.3												
H-2	-26.6	-6.5	-10.1											
ER-1	-13.8	-4.6	5.1	11.4										
ER-2	-3.7	-4.3	-0.8	0.4	1.5									
c1-1	-9.2	-3.6	18.3	21.1	6.9	2.5								
c1-2	-10.6	-5.2	-0.4	3.9	-2.7	0.7	-6.5							
AP-H	-10.0	-4.3	4.9	10.3	1.0	1.7	-4.2	3.1						
AP-ER	-1.4	-2.7	2.5	4.2	1.6	2.2	0.3	2.5	1.3					
AP-c1	0.9	-1.8	8.4	11.1	6.4	3.8	4.0	7.2	5.7	1.7				
ER-c1	-7.47	-5.19	-0.91	2.1	-2.52	0.5	-4.86	-0.5	-2.69	-2.62	-6.43			
H-c1	-20.6	-5.7	-3.6	4.5	-6.6	0.3	-14.6	-1.2	-6.4	-3.2	-9.2	0.3		
H-ER	-13.8	-5.8	-2.6	2.5	-4.9	0.2	-9.3	-1.4	-5.1	-3.2	-8.6	0.6	-0.5	
0,0	-10.1	-3.4	39.5	27.0	9.5	2.6	1.9	7.3	5.5	-0.1	-3.7	5.3	18.5	10.5

In(OR)

	0→H+	0→ER	0→c1	AP→0	AP→H+	AP→ER	AP→c1	c1→H+
0→H+	0.4							
$0 \rightarrow ER$	0.3	na						
0→c1	-0.2	-0.6	-1.0					
AP→0	0.1	-0.1	0.4	-0.7				
AP→H+	0.5	0.3	0.6	0.9	1.3			
$AP \rightarrow ER$	0.5	na	-0.3	0.6	1.2	na		
AP→c1	-0.1	-0.6	-0.8	0.4	0.3	-0.2	na	
c1→H+	0.6	0.9	1.4	0.4	1.0	1.4	1.3	-0.3
ER→H+	0.0	na	0.9	0.2	0.2	na	0.5	-0.4
ER→c1	-0.6	na	-0.5	-0.2	-0.8	na	-0.7	-0.1

z-scores

	0→H+	$0 \rightarrow ER$	0→c1	AP→0	AP→H+	AP→ER	AP→c1	c1→H+
0→H+	4.7							
$0 \rightarrow ER$	2.1	na						
0→c1	-2.2	-2.1	-5.8					
AP→0	1.0	-0.2	0.6	-1.2				
AP→H+	3.4	0.5	2.2	1.4	2.2			
AP→ER	2.5	na	-1.1	0.9	1.5	na		
AP→c1	-0.7	-1.2	-4.3	0.6	0.5	-0.3	na	
c1→H+	5.0	3.0	4.6	1.7	3.6	3.6	4.1	-1.3
ER→H+	-0.3	na	2.5	0.4	0.3	na	1.0	-1.3
ER→c1	-3.4	na	-1.6	-0.5	-1.6	na	-1.2	-0.4

In(OR)

	(AP,CI)	(AP,0)	(0,01)	(AP,H+)	(U,ER)	(0, 11+)	(C1,C1)	(ER,CI)	(H+,C1)	(ER,H+)	(#+,#+)
(0,d1)	na	0.7	0.6	na	0.0	0.6	na	na	0.6	na	0.4
(d1,d1)	na	na	na	na	na	0.9	na	na	na	na	na
(0,d2)	0.0	-0.1	0.2	-0.2	0.3	0.1	0.4	0.0	0.1	0.0	0.2
bioRxiv preprint doi: https://doi.org/10 certified by peer review) is the author/f	1101/603878; th Inder. This article made a	s version posted is a US Governr vailable for use u	April 10, 2019. T nent work, it is no nder a CC0 licen	he copyright hold at subject to copyr se.	er for this preprint ght under 17 US	(which was not C 105 and is also	na	na	0.5	na	0.4
(0,d3)	-0.1	0.4	0.2	0.3	0.4	0.2	0.2	na	0.1	-0.4	0.1
(d3,d3)	-1.1	0.4	0.3	0.3	0.5	0.3	0.5	na	0.5	-0.5	0.1
(0,d1) (0,d2)	na	na	0.2	na	na	0.8	na	na	0.9	na	1.9
(0,d1) (0,d3)	na	na	1.0	na	na	0.6	na	na	1.0	na	0.8
(d1,d1) (0,d3)	na	na	na	na	na	1.6	na	na	na	na	na
(0,d1) (d3,d3)	na	na	na	na	na	1.3	na	na	na	na	na
(0,d2) (0,d3)	-0.1	0.0	0.3	0.0	0.6	0.3	1.2	na	0.3	-0.5	0.4
(d2,d2) (0,d3)	na	na	0.1	na	0.4	0.8	na	na	0.6	na	0.5
(0,d2) (d3,d3)	na	0.2	0.4	na	1.1	0.4	na	na	0.7	-0.1	0.2
(d2,d2) (d3,d3)	na	na	1.3	na	na	0.8	na	na	na	na	na
(0,d1) (0,d2) (0,d3)	na	na	1.4	na	na	1.0	na	na	1.3	na	na
(0,d1) (0,d2) (d3,d3)	na	na	na	na	na	1.5	na	na	na	na	na

(AP,c1) (AP,0) (0,c1) (AP,H+) (0,ER) (0,H+) (c1,c1) (ER,c1) (H+,c1) (ER,H+)(H+,H+)

z-scores

(AP,c1) (AP,0) (0,c1) (AP,H+) (0,ER) (0,H+) (c1,c1) (ER,c1) (H+,c1) (ER,H+)(H+,H+)

-			1	, , ,					1 1		
(0,d1)	na	2.0	2.4	na	-0.1	4.7	na	na	1.7	na	1.1
(d1,d1)	na	na	na	na	na	2.8	na	na	na	na	na
(0,d2)	0.0	-0.7	1.8	-0.7	2.0	1.5	1.2	0.1	0.3	-0.1	1.5
(d2,d2)	na	1.7	2.0	-0.3	0.8	5.5	na	na	1.1	na	1.1
(0,d3)	-0.1	-0.5	2.1	1.3	2.3	3.5	0.7	na	1.0	-1.2	1.0
(d3,d3)	-1.1	2.4	2.1	1.2	2.4	4.3	0.9	na	2.4	-1.3	0.3
(0,d1) (0,d2)	na	na	0.5	na	na	4.1	na	na	1.5	na	1.9
(0,d1) (0,d3)	na	na	2.9	na	na	3.1	na	na	1.8	na	1.4
(d1,d1) (0,d3)	na	na	na	na	na	2.4	na	na	na	na	na
(0,d1) (d3,d3)	na	na	na	na	na	4.5	na	na	na	na	na
(0,d2) (0,d3)	-0.2	-0.5	2.7	0.0	2.5	3.4	2.5	na	1.3	-1.1	1.8
(d2,d2) (0,d3)	na	na	0.5	na	0.7	4.7	na	na	1.2	na	1.2
(0,d2) (d3,d3)	na	0.8	2.0	na	3.2	3.9	na	na	2.1	-1.1	0.8
(d2,d2) (d3,d3)	na	na	3.0	na	na	3.0	na	na	na	na	na
(0,d1) (0,d2) (0,d3)	na	na	2.9	na	na	3.2	na	na	1.6	na	na
(0,d1) (0,d2) (d3,d3)	na	na	na	na	na	3.0	na	na	na	na	na

MHC Genotype	RR ₁	RR ₂	Additive Model	Observed	Multiplicative Model
AP , AP	1	1	1	1	1
AP , 0	1	4	4	4	4
0,0	4	4	7	8	16
AP, H+	1	14	14	14	14
0 , H+	4	14	17	23	56
H+, H+	14	14	27	49	196
AP, ER	1	8	8	8	8
0 , ER	4	8	11	15	32
ER , H+	8	14	21	33	112
AP , c1	1	3	3	3	3
c1 , c1	3	3	5	25	9
0, c1	4	3	6	8	12
c1 , H+	3	14	16	31	42
c1 , ER	3	8	10	29	24

2-Genotype	RR.	RR.	Additive	Observed	Multiplicative	
Combinations	$\Lambda\Lambda_I$	$\pi\pi_2$	Model	Observed	Model	
(0, 0) + (0, d1)		15		15		
(0, 0) + (d1, d1)		20		20		
(0, 0) + (0, d2)		8		8		
(0, 0) + (d2, d2)		10		10		
(0, 0) + (0, d3)		9		9		
(0, 0) + (d3, d3)		11		11		
(0, H+) + (0, d1)	23	15	37	42	345	
(0, H+) + (d1, d1)	23	20	42	59	460	
(0, H+) + (0, d2)	23	8	30	25	184	
$(0, H^+) + (d2, d2)$	23	10	32	42	230	
(0, H+) + (0, d3)	23	9	31	28	207	
$(0, H^+) + (d3, d3)$	23	11	33	31	253	
(H+, H+) + (0, d1)	49	15	63	69	735	
$(H^+, H^+) + (0, d2)$	49	8	56	57	392	
(H+, H+) + (d2, d2)	49	10	58	67	490	
(borxiv preplint doi: https://dd.org/b3101/60387	B; this version pos	red April 90, 2019.	The copyright holder for this p	eprint (which was not	441	
(H+H+) + (d3, d3)	rticle is a US Cove de available for us	rnment work. It is e under a CC0 lic	hot subject to copyright under ense. 59	7 USC 105 and is also 54	539	
(0, cl) + (0, dl)	3	15	17	15	45	
(0, c1) + (0, d2)	3	8	10	10	24	
(0, c1) + (d2, d2)	3	10	12	12	30	
(0, c1) + (0, d3)	3	9	11	10	27	
(0, c1) + (d3, d3)	3	11	13	11	33	
(c1, c1) + (0, d2)	25	8	32	36	200	
(c1, c1) + (0, d3)	25	9	33	30	225	
(c1, c1) + (d3, d3)	25	11	35	39	275	
$(H^+, c1) + (0, d1)$	31	15	45	57	465	
(H+,c1) + (0,d2)	31	8	38	31	248	
(H+,c1) + (d2, d2)	31	10	40	46	310	
$(H^+, c_1) + (0, d_3)$	31	9	39	35	279	
(H+,c1) + (d3, d3)	31	11	41	51	451	
(AP = 0) + (0 = d1)	4	15	18	8	60	
(AP = 0) + (0 = d2)	4	8	10	4	32	
$(AP \ 0) + (0 \ d2)$	4	8	13	7	40	
(AP 0) + (0, d2)	4	9	12	4	36	
$(AP \ 0) + (d3 \ d3)$	4	11	14	7	44	
(AP H+) + (0 d3)	14	9	22	17	126	
(AP H+) + (d3 d3)	14	11	24	18	154	
(AP c1) + (0 d2)	3	8	10	3	24	
(AP c1) + (0, d2)	3	0	10	3	27	
(AP c1) + (d3 d3)	3	11	13	1	33	
$(0 \ ER) + (0 \ d1)$	15	15	29	16	225	
(0, ER) + (0, d2)	15	15	22	10	120	
(0, ER) + (0, a2)	15	0	24	20	120	
$(0, EI) + (a_2, a_2)$	15	0	24	30	130	
(0, EI) + (0, a3)	15	9	25	20	155	
(U, ER) + (a3, a3) (EP H+) + (0 - 12)	22	0	40	23	264	
$(ER, H^{+}) + (0, d2)$	22	0	40	25	204	
$(ER, H^+) + (0, d3)$	22	11	41	33	262	
$(ER, \Pi^{+})^{+} (a3, a3)$	33	0	45	20	264	
(ER, CI) + (0, a2)	55	0	40	29	204	

3-Genotype Combinations	RR ₁	RR ₂	RR ₃	Additive Model	Observed	Multiplicative Model
(0, 0) + (0, d1) + (0, d2)	8	15	8	29	14	960
(0, 0) + (d1, d1) + (0, d2)	8	10	8	24	24	640
(0, 0) + (0, d1) + (d2, d2)	8	15	10	31	20	1,200
(0, 0) + (0, d1) + (0, d3)	8	15	9	30	20	540
(0, 0) + (d1, d1) + (0, d3)	8	20	9	35	19	720
(0, 0) + (0, d1) + (d3, d3)	8	15	11	32	15	660
(0, 0) + (0, d2) + (0, d3)	8	8	9	25	10	576
(0, 0) + (d2, d2) + (0, d3)	8	10	9	30	11	720
(0, 0) + (0, d2) + (d3, d3)	8	8	11	31	12	704
(0, 0) + (d2, d2) + (d3, d3)	8	10	11	27	14	600
(AP, 0) + (0, d1) + (0, d3)	4	15	9	26	11	540
(AP, 0) + (0, d2) + (0, d3)	4	8	9	19	5	288
(AP, 0) + (d2, d2) + (0, d3)	4	10	9	21	7	360
bioRxiv preprint doi: https://doi.org/10.1101/603878; this version posted certified by peer review) is the author/funder. This article is a WS Govern made available for use	a April 10, 2019. ment work. It is n under a CC0 lice	he copyright hole ot subject to copy nse.	er for this preprin right under 17 US	t (which was not C 105 and is also	6	352
(0, c1) + (0, d1) + (0, d2)	8	15	8	29	10	960
(0, c1) + (0, d1) + (0, d3)	8	15	9	30	22	1,080
(0, c1) + (0, d2) + (0, d3)	8	8	9	23	10	576
(0, c1) + (d2, d2) + (0, d3)	8	10	9	35	12	720
(0, c1) + (d2, d2) + (d3, d3)	8	10	11	27	31	880
(0, ER) + (0, d2) + (0, d3)	15	8	9	30	23	1,080
(0, ER) + (d2, d2) + (0, d3)	15	10	9	42	19	1,350
(0, ER) + (0, d2) + (d3, d3)	15	8	11	32	38	1,320
(0, H+) + (0, d1) + (0, d2)	23	15	8	44	51	2,760
(0, H+) + (d1, d1) + (0, d2)	23	20	8	49	44	3,680
(0, H+) + (0, d1) + (d2, d2)	23	15	10	46	45	3,450
(0, H+) + (0, d1) + (0, d3)	23	15	9	45	39	3,105
(0, H+) + (d1, d1) + (0, d3)	23	20	9	50	85	4,140
(0, H+) + (0, d1) + (d3, d3)	23	15	11	47	107	3,795
(0, H+) + (0, d2) + (0, d3)	23	8	9	38	31	1,656
(0, H+) + (d2, d2) + (0, d3)	23	10	9	39	50	2,070
(0, H+) + (0, d2) + (d3, d3)	23	8	11	40	35	2.024
(0, H+) + (d2, d2) + (d3, d3)	23	10	11	42	48	2,530
(AP, H+) + (0, d2) + (0, d3)	23	15	9	40	15	2,024
(AP, H+) + (0, d2) + (d3, d3)	23	15	11	47	26	3,795
(ER, H+) + (0, d2) + (0, d3)	33	8	9	48	28	2,376
(ER, H+) + (0, d2) + (d3, d3)	33	8	11	50	46	2,904
(H+, c1) + (d2, d2) + (0, d3)	31	10	9	48	55	2,790
(H+, c1) + (0, d2) + (d3, d3)	31	8	11	48	60	2,728
(H+, H+) + (0, d1) + (0, d2)	49	15	8	70	350	5,880
(H+, H+) + (0, d1) + (0, d3)	49	15	9	72	108	6,615
(H+, H+) + (0, d2) + (0, d3)	49	8	9	64	75	3,528
(H+, H+) + (d2, d2) + (0, d3)	49	10	9	66	84	4,140
(H+, H+) + (0, d2) + (d3, d3)	49	8	11	66	63	4,312

4-Genotype Combinations		RR ₂	RR3	RR₄	Additive Model	Observed	Multiplicative Model
(0, 0) + (0, d1) + (0, d2) + (0, d3)	8	15	8	9	37	20	8,640
(0, 0) + (0, d1) + (0, d2) + (d3, d3)	8	15	8	11	39	16	10,560
(0, H+) + (0, d1) + (0, d2) + (d3, d3)	23	15	8	9	52	59	24,840
(0, c1) + (0, d1) + (0, d2) + (0, d3)	8	15	8	9	37	22	8,640